

P

30

UR

EUI WORKING PAPERS IN ECONOMICS

EUI Working Paper ECO No. 91/59

Reconciling the Term Structure of Interest Rates with the Consumption Based ICAP Model

> FABIO CANOVA and JANE MARRINAN

ean University Institute, Florence



EUROPEAN UNIVERSITY INSTITUTE, FLORENCE

ECONOMICS DEPARTMENT

EUI Working Paper ECO No. 91/59

Reconciling the Term Structure of Interest Rates with the Consumption Based ICAP Model

FABIO CANOVA and JANE MARRINAN

BADIA FIESOLANA, SAN DOMENICO (FI)

All rights reserved. No part of this paper may be reproduced in any form without permission of the authors.

> © Fabio Canova and Jane Marrinan Printed in Italy in December 1991 European University Institute Badia Fiesolana I-50016 San Domenico (FI) Italy

Reconciling the Term Structure of Interest Rates with the Consumption Based ICAP Model

Fabio Canova *

Department of Economics, Brown University, Providence, RI, 02912 and Department of Economics, European University Institute I-50016 San Domenico di Fiesole, Firenze, Italy Canova@IFIIUE.bitnet Telephone: 39-55-5092-252; Fax: 39-55-5092-202

and

Jane Marrinan Department of Economics, Boston College, Chestnut Hill, MA 02167 and Department of Economics, European University Institute Marrinan@IFIIUE.bitnet

> First Draft: October 1990 This version: September, 1991

Abstract

This paper attempts to explain some of the time series features of the low end of the term structure of US interest rates using a representative agent cash-in-advance consumption based ICAP model, modified to allow for time variation in the conditional variances of the exogenous processes. The ability of the model to reproduce features of the actual data is evaluated using a Monte Carlo simulation technique. The statistical properties of simulated yields and spreads are shown to replicate important properties of the observed term structure of U.S. T-bills over the sample 1964-1988.

JEL Classification No.: 212, 313.

Key Words: Yields, Spreads, Heteroschedasticity, Cash-in-advance, Simulation.

*We have benefitted from conversations and comments of a number of people including David Backus, Simon Benninga, Tom Cooley, Wouter Den Haan, Roger Farmer, Marc Gertler, Bruce Hansen, Eric Jacquier, Bob King, Adrian Pagan, Franco Peracchi, Danny Quah, Mark Salmon, Ken Singleton, Gregor Smith, Alan Stockman, George Tauchen and Stan Zin. We would also like to thank the participants of seminars at Brown, European University Institute, NYU, Montreal, Rochester, Society for Economic Dynamics and Control Meetings 1991, Capri; North American Summer Meetings of the Econometric Society 1991, Philadephia; NBER summer meetings 1991, European Finance Association Meetings 1991, Rotterdam; International Conference in Finance (AFFI) 1991, Louvain and European Meetings of the Econometric Society 1991, Cambridge 1991 for useful suggestion.



1 Introduction

One of the most widely examined theories in financial economics is the expectations theory of the term structure of interest rates, which relates the yields on long term bonds to expected future yields on short term bonds. The theory has been tested in several ways. Shiller (1979), Campbell (1986), Campbell and Shiller (1984, 1987, 1990), Mankiw (1986) and others have used regression tests to examine whether the slope of the yield curve (the spread) has predictive content for the holding premium return. Startz (1982), Fama (1984), Fama and Bliss (1987), Stambaugh (1988) and others have tested the theory using the predictive content of forward rates for realized future spot interest rates. To provide a more direct test of the predictive content of forward rates Froot (1989) uses survey data on expectations of future spot interest rates. Much of the empirical evidence presented so far indicates that the theory has severe shortcomings. The yield spread has information in predicting holding premium returns and the forward rate does not accurately predict future spot interest rates. Standard explanations for these failures, which include the presence of time varying risk and liquidity premia or irrational behavior, have yet to provide a convincing account of the empirical features of the term structure of interest rates.

One recent line of research has examined whether this empirical evidence is consistent with the consumption based theory of risk premia developed by Lucas (1978) and Breeden (1979) and implicit in the one sector growth model (see e.g. Backus, Gregory, Zin (1989), Donaldson, Johnsen and Merha (1990) and Rouwenhorst (1990)). This research has attempted to determine whether numerical versions of the theory can account for variation in risk premia which are implicit in the failure of the expectations hypothesis. Some success has been reported in matching the variability of yields (see Den Haan (1990)), the behavior of real and nominal yields over the business cycle (Donaldson, Johnsen and Merha (1990) and Labadie (1991)), and the predictive power of the nominal yield curve for real output (Cooley and Ohanian (1990)).

This paper contributes to this growing body of literature in several ways. As in the above studies we are interested in analyzing whether modifications of a standard consumption based intertemporal asset pricing model can generate a term structure of interest rates which is consistent with US empirical evidence. We differ from the others in two ways. First, we allow the conditional variability of the exogenous forces of the economy to vary over time. We want to determine whether changes over time in the uncertainty surrounding economic fundamentals can help to quantitatively explain the behavior of the yield curve (see Breeden (1986) and Stambaugh (1988)). Second, we use a Monte Carlo simulation technique to assess the ability of the model to reproduce features of the actual data.

In comparing actual and simulated data we focus on a broad set of features of the actual yield curve. They are: (i) the yield curve is upward sloping on average, (ii) the volatility of yields decreases with maturity, (iii) yields at all maturities are highly autocorrelated and highly heteroschedastic, (iv) changes in yields on longer term bills have significant correlation with changes in yields on shorter term bills, (v) the yield spread is a better predictor for future changes in short term yields than the forward premium. The opposite is true for future changes in longer term yields.

To attempt to replicate these features quantitatively we simulate a monetary model with time separable preferences and exogenous endowments, fiscal policies and monetary policies. Under general conditions we show that the equilibrium interest rates depend, among other things, on the conditional variability of the exogenous forces of the economy (see also Longstaff and Schwartz (1990)). In a related paper (Canova and Marrinan (1990)) we demonstrated that by allowing for time variation in the conditional distributions, a two country version of the model is able to quantitatively replicate the variability, serial correlation properties and heteroschedastic structure of profits from forward speculation in foreign exchange markets. Work by Ferson and Harvey (1991) also indicates that changes in the uncertainty surrounding economic variables may be important in characterizing properties of US stock returns. Therefore, it is of interest to examine whether a unified explanation for the apparent failure of the expectations theory in financial markets is possible.

To asses the quantitative properties of the theoretical economy we use a Monte Carlo methodol ogy (see Canova (1990) for a complete description of the technique). The approach is advantageous in several respects. It includes both estimation by simulation and calibration techniques as special cases. It also allows us to incorporate existing evidence on the parameters of the model in a realistic way, make probability statements on the range of possible outcomes that the model can generate or on particular events we are interested in replicating and, as a by-product, provides a global sensitivity analysis for some crucial parameters. We take the economic model to be, at best, an approximation to reality and we view ourselves as trying to determine how good an approximation it is. This is done by taking the actual realization of the statistic of interest as a critical value and computing the probability (over replications) that the model generates that critical value. Gregory and Smith (1991) have proposed a similar approach to formalize inference in simulated macroeconomic models. Contrary to their approach, we explicitly take parameter uncertainty into consideration and randomize over *both* exogenous processes *and* parameters (see also Kwan (1990)). The parameters are drawn from the frequency distribution of the estimates of the parameters compiled from results existing in the literature.

The results indicate that the model can, with high probability, reproduce several interesting aspects of the term structure, namely, the upward sloping average yield curve, the fact that volatility of yields decreases with maturity and most of the second order properties of the spreads. The model falls somewhat short in quantitatively accounting for other features of the term structure, including the serial correlation properties of yields and their heteroskedastic structure and, partially, the cross correlations of long and short yields, of the spreads and forward premia with changes in yields.

We find that the presence of heteroskedasticity in the exogenous processes is fundamental in making simulated yields comparable to actual yields. With homoschedastic exogenous processes the second order properties of simulated yields are with probability 1 at odds with actual data. We also find that allowing for a structural break in the stochastic process of the money supply enhances the ability of the model to match the serial correlation and the heteroskedastic properties of the data.

The paper is organized in seven sections. The next section presents evidence on the short end of the term structure of US interest rates and establishes some stylized "facts". Section 3 describes the model economy and derives equilibrium pricing formulas for nominally risk free interest rates. Section 4 presents our simulation technique and the relationship with existing simulation approaches. Section 5 presents the results. Section 6 discusses the sensitivity of the results to several modifications of the basic model. Section 7 concludes.

2 Evidence from the Term Structure of U.S. T-bill Rates

This section presents some "facts" concerning the short end of term structure of US interest rates which emerge from the available data set. We concentrate on the short end of the term structure because we are primarily interested in analyzing how time variation in the conditional variances of economic fundamentals affects yields. At very long horizons agents' forecasts of the conditional variances of fundamentals are likely to coincide with the unconditional variances unless their conditional distributions display very strong persistence. In addition, by limiting the scope of the research to the short end of the term structure, we avoid having to deal with possible "preferred habitat" considerations which may require more complicated theoretical settings. We present simple summary statistics instead of regression coefficients because they are more robust to interpolation and measurement errors and to small sample biases.

The data we employ is obtained from the Center for Research in Security Prices (CRSP) tapes, augmented with Fama's term structure files on T-bills (Fama (1984)). Monthly data for the average of the bid and ask for spot and forward prices of 1-3-6-12 month T-bills are taken from Fama's 12 month bill files. Yields are continuously compounded over a month and converted to percentages per year by multiplying the figures by 1200 to express the data in more familiar units. For each, month Fama's data set chooses the bill with maturity closest to 12-months. This bill is then followed to maturity, providing in subsequent months the yields for maturities of 11 months, 10 months, etc. Since data for 12 month T-bills is available from 1963,7, data for a 1-month bill is available from 1964,6 and in our study we use data up to 1987,11 for a total of 281 observations ¹. Figures 1 and 2 present the time plots for the yields and the spreads and their estimated univariate $M\bar{A}$ representations. Tables 1 and 2 report selected summary statistics and table 3 some relevant cross moments among the variables. Since the yield plots display a break around 1979, we also compute statistics for two subsamples 64,6-79,9 and 79,10-87,11. Although the magnitudes of the first two moments of yields change across subsamples, neither the serial correlation properties nor any of the qualitative aspects of summary statistics presented are altered across subsamples. Therefore The / we present only evidence concerning the entire sample 2 .

The evidence contained in the tables and figures can be summarized as follows:

• The arithmetic mean of nominal yields on all T-bills for the 1964-1987 period is close to seven

¹Since the data set on 12-month prices contains several missing values, we reconstruct missing values either by linear interpolation or by compounding the prices of bills of various maturities. The statistics we report are insensitive to which of the two procedures are used and are practically identical to those obtained by simply dropping missing values from the sample. For the sake of robustness we also examined the properties of the data set employed by McCulloch and Shiller (1987) where missing observations are interpolated with a cubic spline and the term structure data contained in the Citibase Tape, where there are no missing observations but data is reported as average over the month. We find that, apart from very minor numerical differences, the qualitative features of the statistics reported are robust. Regression coefficients, however, do differ substantially depending on the data set used.

²Results for subsamples are contained in an appendix available from the authors on request. Note that the spreads appear to be stationary throughout the sample.

percent. The average term structure is slightly upward sloping (see Fama (1984)) but average yields do not increase proportionally with the gap in maturities.

- The standard error of yields averages about 2.8 but over the term structure is slightly humpshaped, with the hump occurring for the 3-month maturity. Volatility, defined as the standard error of the series divided by the absolute value of the mean, is slightly decreasing with maturity. This is in contrast with some existing empirical findings comparing bonds of long maturities and T-bills (see e.g. Shiller (1979) or Schotman (1991)).
- Except for the 12 month bills, yields are skewed to the left (lower than average yields occur more frequently than higher than average yields) and highly leptokurtic. The Kendall and Stuart (1958) two-tailed test rejects the null hypothesis that their conditional distribution is normal for all maturities. The spreads between yields of different maturities also display marked nonnormalities. This behavior persists when the gap between maturities increases. Therefore, contrary to what has been found in other financial markets (see e.g. Diebold (1988) and Fama (1976)), time aggregation does not reduce nonnormalities.
- Yields at all maturities are highly serially correlated (see Fama (1984)) and their univariate moving average representation decays slowly. A 1% shock still generates a .20% displacement in the level of yields at the 48 month horizon. The degree of persistence in the autocovariance function increases with maturity. This is consistent with the idea that yields on longer term instruments reflect events in the future which are unaffected by current business cycle conditions (see e.g. Donaldson, Johnsen and Merha (1990)). Changes in T-bill yields of any maturity have a very small and insignificant first order autocorrelation coefficient. However, except for 1 month bills and contrary to Fama (1984), we reject the hypothesis that yield changes are white noise because of significant seasonalities present in the data (see the surge in the MA representation of 3-6-12 month yields at the 12 month horizon).
- The conditional distribution of the yields and spreads display time variation and marked nonlinearities. The conditional variances of yields of all maturities are highly volatile and display significant conditional heteroskedasticity (see also e.g. Campbell (1987) and Engle, Lilien and Robbins (1987)). The nature of the heteroskedasticity in yields varies across maturities since spreads still display marked heteroskedastic patterns.

- There is a negative contemporaneous correlation between the forward premia (the difference between the forward and the spot rate) and interest rate changes (on average, -0.15) which is inconsistent with the predictions of the liquidity theory of the yield curve (see e.g. Kessel (1965)). The spread appears to have higher predictive content for future changes in short term interest rates than the forward premium, but the difference is small (compare with Fama (1984)). For future changes in longer term rates the opposite is true.
- The contemporaneous correlation between changes in long term yields and changes in short term yields averages around .70 and slightly decreases as the gap between maturities increases. This result, taken together with the flat pattern of standard errors at different maturities, is consistent with Campbell and Shiller (1984) and Mankiw and Summers (1984) and with the idea of "undersensitivity" of longer rates to short rate fluctuations.

idea of "undersensitivity" of longer rates to short rate fluctuations.

Model-based empirical work of the term structure has generally been concerned with the estimation of the risk aversion parameter and of the discount factor of the representative consumer (see e.g. Hansen and Singleton (1983), Brown and Gibbons (1985), Dunn and Singleton (1986), Singleton (1989) or Lee (1989)). It is only more recently, with the work of Backus, Gregory and Zin (1989), Rouwenhorst (1990), Merha, Johnsen and Donaldson (1990), Cooley and Ohanian (1990), Den Haan (1990), and Labadie (1991) that the emphasis has been shifted to try to ascertain whether a consumption based ICAP model can reproduce those features of the US yield curve which are puzzling from the point of view of the expectations theory. Our work is a direct extension of their efforts.

3 A General Equilibrium Model of the Term Structure

The theoretical framework we employ is a version of the cash-in-advance monetary model developed by Lucas (1980), modified to allow for time variation in the conditional variance of the exogenous processes. It is similar to the one used in a previous paper of ours (Canova and Marrinan (1990)) in which we study the behavior of profit from forward speculation in foreign exchange markets. We employ a similar theoretical structure because we are interested in assessing whether such a model can quantitatively account for a wide variety of features of US financial markets. Since the model is well known in the literature, we only briefly describe its features and proceed directly to the computation of the equilibrium values of the variables of interest.

Every period the economy is endowed with Y_t units of a nonstorable consumption good. There is a government which consumes G_t units of the good. To finance these consumption requirements the government issues money, M_t , collects real lump sum taxes, T_t , and issues debt to finance any purchases in excess of money creation and tax collections. This debt is in the form of state contingent nominal bills of maturity k, k = 1, 2, ..., K. Endowments, government consumption requirements and money supplies are exogenous and follow a first order Markov process with stationary and ergodic transition function.

The economy is populated by a representative household maximizing a time separable utility function. The household is subject to both a wealth constraint and a liquidity constraint which compels it to purchase goods with cash. The timing of the model follows Lucas with asset markets open first and goods markets following. At the beginning of each period the consumer enters the asset market and decides how to allocate her wealth among the productive assets, currency, and the state contingent nominal bonds. After the asset market closes, the consumer enters the goods market and makes her consumption purchases with previously accumulated currency.

Equilibrium requires that households optimize and all markets clear. Since capital markets are complete, this permits an unconstrained Pareto optimal allocation of the time-state contingent nominal bonds. Let $e^{-r_{t,k}(v)}$ denote the discount price of a bill paying 1 unit of currency at time t + k, if event v occurs and $r_{t,k}(v)$ denote the associated continuously compounded interest rate. By integrating the equilibrium pricing formulas over all possible v we can determine the price at t of a nominally riskless k period discount bill, $e^{-r_{t,k}}$.

In equilibrium nominal interest rates reflect optimal consumption-saving decisions by equating bond prices to individuals' expected marginal rate of substitution of future nominal expenditure for current nominal expenditure

$$e^{-r_{t,k}} = \beta^k E_t \frac{P_t U_{t+k}(C_{t+k})}{P_{t+k} U_t(C_t)}$$
(1)

Because of the timing of the model, all uncertainty is resolved prior to the household's money holding decisions so they hold just enough currency to finance their current consumption purchases.

)

This implies that $P_t = \frac{M_t}{Y_t}$ and the discount price of a bill of maturity k is:

$$e^{-\tau_{t,k}} = \frac{\beta^k E_t M_{t+k}^{-1} Y_{t+k} U_{t+k}(C_{t+k})}{Y_t M_t^{-1} U_t(C_t)}$$
(2)

From (2) is it immediate to compute forward prices for maturity q, $f_{t,q}$ as:

$$e^{-f_{t,q}} = \frac{e^{-r_{t,k+q}}}{e^{-r_{t,k}}} \quad \forall k,q$$
 (3)

Yields and forward rates can be obtained from (2) and (3) by simple logarithmic transformations.

An expression for the slope of the yield curve (the spread) between k and h-period nominally riskless pure discount bills with $k > h \ge 1$ is obtained from (2) as:

$$SP_t^{k,h} = h^{-1} \ln[\frac{E_t \beta^h Y_{t+h} (M_{t+h})^{-1} U_{t+h}}{Y_t (M_t)^{-1} U_t}] - k^{-1} \ln[\frac{E_t \beta^k Y_{t+k} (M_{t+k})^{-1} U_{t+k}}{Y_t (M_t)^{-1} U_t}])$$
(4)

Finally, an expression for the forward premium, defined as $FP_t^{q,h} = -\ln(e^{-f_{t,q}}) + \ln(e^{-\tau_{t,h}})$, is:

ally, an expression for the forward premium, defined as
$$FP_t^{q,h} = -\ln(e^{-f_{t,q}}) + \ln(e^{-r_{t,h}})$$
, is:

$$FP_t^{q,h} = (k+q)^{-1}\ln[\frac{E_t\beta^{k+q}Y_{t+k+q}(M_{t+k+q})^{-1}U_{t+k+q}}{Y_t(M_t)^{-1}U_t}]) - k^{-1}\ln[\frac{E_t\beta^kY_{t+k}(M_{t+k})^{-1}U_{t+k}}{Y_t(M_t)^{-1}U_t}]]$$

$$- h^{-1}\ln[\frac{E_t\beta^hY_{t+h}(M_{t+h})^{-1}U_{t+h}}{Y_t(M_t)^{-1}U_t}]])$$
(5)

Yields, forward rates, spreads and forward premia depend on expectations about future output future money supply and future consumption growth. Since in equilibrium expectations about $\overset{\circ}{\ominus}$ future consumption growth depend on expectations about future government purchases of goodsboth supply and demand factors affect the position and the slope of the term structure. Also $\frac{1}{20}$ uncertainty about regime shifts or regime persistence can influence the expectation formation and therefore the properties of forward and spot rates.

To obtain closed form expressions for yields, forward rates, spreads and forward premia the instantaneous utility function is specialized to be of a constant relative risk aversion type as:

$$U(C_t) = \frac{C_t^{(1-\gamma)}}{1-\gamma} \quad 0 \le \gamma < \infty \tag{6}$$

where γ is the parameter of risk aversion. Let Φ_t be the proportion of government consumption in total output at time t. In equilibrium $C_t = Y_t - G_t = Y_t(1 - \Phi_t)$. Evaluating the marginal utilities in (2)-(5) at these equilibrium consumption levels gives expressions for yields and forward rates entirely in terms of the distributions of the exogenous variables. The complete solution requires substituting in specific processes governing the exogenous variables.

Let $z_t = [\Delta \log(Y_t), \Delta \log(M_t), \Phi_t]$. We assume that z_t has a stationary unconditional distribution. In addition, we assume that all three processes follow a first order autoregression

$$z_{jt} = A_{0j} + A_{1j} z_{jt-1} + \epsilon_{jt} \quad j = 1, \dots, 3$$
(7)

and that their conditional variances are time varying and follow a GARCH(1,1) process:

$$\sigma_{jt}^2 = a_{0j} + a_{1j}\sigma_{jt-1}^2 + a_{2j}\epsilon_{jt-1}^2, \quad j = 1, \dots, 3$$
(8)

If, as in Breeden (1986), we take a second order Taylor expansion of (2)-(5) around z_t , it is immediate to show that yields, forward rates, spreads and forward premia will all depend on the conditional means, variances and covariances of the exogenous processes. Since there is evidence that the conditional covariances are small (see e.g. Hansen and Hodrick (1983) and Braun (1990)) ³, we will include them along with the higher order terms in the approximation error and neglect them in the simulations. This allows us to focus on the contribution of time varying conditional variances to the properties of the term structure. Closed form expressions for the four variables of interest appear in the appendix.

Straightforward calculations indicate that:

- the unconditional means and variances of the exogenous processes influence the average size of all four variables.
- deviations of the conditional moments relative to the unconditional moments of the exogenous processes affect the unconditional autocovariance functions of all four variables.
- the discount factor β affects only the mean of yields.
- The risk aversion parameter, γ, affects both the unconditional means and the unconditional autocovariances of all four variables.

Since (2)-(5) hold for each k, it is possible to express long term rates, spreads and forward premia as a function of the distributional characteristics of short term rates, using the approach suggested by Longstaff and Schwartz (1990). In particular, the term structure of yields for maturities greater

³Using Citibase data we estimated the contemporaneous correlation of industrial production growth and monetary base growth at a monthly frequency to be 0.04 with a standard deviation of 0.07. The contemporaneous correlations between government expenditure and output and government expenditure and monetary base at a quarterly frequency were insignificantly different from zero as well.

than 1 month will depend on the level of the 1 month yield and on a few terms of its conditional autocovariance function. Longstaff and Schwartz demonstrated that the fit of their model for yields of long maturities improves when the conditional variance of short term yields appears as a regressand in addition to the level of short term yields. Our model implies that, in addition to these two factors, time variations in the autocovariance function of short term yields is important in explaining movements in the term structure ⁴.

4 Simulating the model

To generate time series for the variables of interest, it is necessary to select values for the 17×1 vector of parameters $\theta = (\gamma, \beta, A_{01}, A_{11}, a_{01}, a_{11}, a_{21}, A_{02}, A_{12}, a_{02}, a_{12}, a_{22}, A_{03}, A_{13}, a_{03}, a_{13}, a_{23})$. To provide discipline in the simulation one could, as in "calibration" exercises, select them to be consistent with existing micro studies. Alternatively, one could estimate θ by simulation. That is, one could choose θ to formally match statistics of the simulated and of the actual data in the least squares metric (see Lee and Ingram (1990); Duffie and Singleton (1990)) or in the VAR metric (see Smith (1990)).

The approach we employ here incorporates ideas of Monte Carlo testing developed in Marriott (1979) and has several appealing features (see Canova (1990) for a complete description of the methodology). It allows us to summarize existing econometric evidence on the parameters in a realistic way, automatically provides a global sensitivity analysis for reasonable perturbations of the parameters and permits a more formal evaluation of the properties of the model.

Our task is to generate probability statements for statistics of the simulated data. For example, we would like to know what is the probability that the model can generate, on average, an upward sloping yield curve. Available information on the parameters is summarized by means of a joint of density $\pi(\theta|\mathcal{F})$, where \mathcal{F} is the information set available and $\theta \in \Theta \subset \mathbb{R}^{17}$. Let $G(x_t(z_t)|\theta, m)$ be the density for the $q \times 1$ vector of endogenous time series x_t , conditional on the parameter vector θ and the particular economic model m we have chosen. Here x_t includes four yields, six spreads and six forward premia. $G(x_t(z_t)|\theta, m)$ describes the likelihood of obtaining an x_t path from our model once a particular θ vector is chosen. For given θ , randomness in x_t is due to the randomness

⁴Using the instrumental variable procedure suggested by Pagan and Ullah (1988) we find that the first two conditional autocovariance terms of 1 month yields enter significantly in a regression of longer term yields on the level of 1 month yields, on its conditional variance and on 4 conditional autocovariances.

in the exogenous processes z_t .

Let $J(x_t(z_t), \theta | m, \mathcal{F})$ be the joint distribution of x_t and θ given the model specification and the information set. In the analysis we focus on statistics of the simulated data which are functions $h(\theta, z_t)$ of the parameters θ and of the exogenous processes z_t . In our case $h(\theta, z_t)$ includes the first and second conditional moments, the first four unconditional moments, terms of the autocovariance function of x_t and some cross correlations among its elements. Model based probabilities for $h(\theta, z_t)$ can be obtained for any $\mathcal{A} \subset \Theta$ by evaluating integrals of the form:

$$E(h(\theta, z_t)|m, \mathcal{F}, \mathcal{A}) = \int_{\mathcal{A}} h(\theta, z_t) J(x_t(z_t), \theta | m, \mathcal{F}) d\theta dz_t$$
(9)

Although theoretically straightforward, expressions like (9) are generally impossible to compute analytically or using simple numerical (spherical or quadrature) rules when Θ is high dimensional. Our approach is to use a Monte Carlo methodology ⁵. The main idea is simple. Let θ_i be a $k \times 1$ dimensional i.i.d. vector of parameters and $\{z_{it}\}_{t=1}^{T}$ be a path for z_t where the subscript *i* refers to the draw. If the probability function from which the θ 's and the *z*'s are drawn is proportional to $J(x(z_t), \theta | m, \mathcal{F}_t)$, then, by the law of large numbers, $n^{-1} \sum_{i=1}^{n} h(\theta_i, z_{it})$ converges almost surely to $E(h(\theta, z_t))$, where *n* is the number of replications. Therefore, by drawing a large number of replications for θ and *z* from $J(x_t(z_t), \theta | m, \mathcal{F}_t)$, we can approximate arbitrarily well $E[h(\theta, z_t)]^6$.

This Monte Carlo approach to simulation explicitly accounts for the uncertainty faced by a simulator in choosing parameter values and encompasses both calibration and estimation by simulation as special cases. Calibration is obtained when $\pi(\theta|\mathcal{F})$ has a point mass at a given $\bar{\theta}$ (usually chosen on the basis of micro-studies) and when a single draw from $G(x_t(z_t)|\bar{\theta}, m)$ is made. Some authors report results when outcomes are averaged over a small number of simulations (see e.g. Backus, Gregory and Zin (1989)). In this case, $\pi(\theta|\mathcal{F})$ still has a point mass at $\bar{\theta}$ but repeated draws from $G(x_t(z_t)|\bar{\theta}, m)$ are made.

The simulated method of moments (SMM) of Lee and Ingram (1990) or Duffie and Singleton (1990) and the GMM procedure of Burnside, Eichenbaum and Rebelo (1990) are also special cases of

⁵See Tanner and Wong (1987), Gelfand and Smith (1990) and Niederreiter (1988) for alternative approaches.

⁶When $G(x_t(z_t)|\theta, m, \mathcal{F})$ is unknown, and numerical procedures are needed to solve the model, one could follow Rubin (1987) and Geweke (1989) and draw from the *Importance Sampling* density of θ and z_t . Under mild regularity conditions the laws of large numbers still apply, i.e.: $\frac{\sum_{j=1}^{n} h(\theta_i, z_t, i), w_i}{\sum_{i=1}^{n} w_i} \equiv h_n \to E(h(\theta, z_t))$ and $\sqrt{n}[h_n - E(h(\theta, z_t))] \Rightarrow$ $\mathcal{N}(0, \sigma_h^2)$ where $w_i = \frac{J(x_t(z_t, i), \theta_i | m, \mathcal{F}_t)}{I(\theta_i, x_t(z_t, i))}, \sigma_h^2 = var(h(\theta))$ and $I(\theta, x_t(z_t))$ is the Importance Sampling density. Geweke (1989) describes how in practice one would select $I(\theta, x(z_t))$. this framework. In both cases $\pi(\theta|\mathcal{F})$ is a density with a point mass at θ^* , where θ^* is either a vector of parameters which minimizes a measure of distance between simulated and actual data or sets some orthogonality conditions equal to zero. Simulations are performed by drawing one or more realizations from $G(x_t(z_t)|\theta^*, m)$. Similarly, the simulated quasi-maximum likelihood technique (SQML) of Smith (1990) obtains when $\pi(\theta|\mathcal{F})$ has a point mass at $\hat{\theta}$, the SQML estimator of θ , and simulations are performed by drawing one or more realizations from $G(x_t(z_t)|\hat{\theta}, m)$.

4.1 Model evaluation

Probability statements and quantiles for the statistics of interest are easily obtained as a by-product of the Monte Carlo procedure. For example, to evaluate $P(h(\theta, z_t) \in A)$, where A is a bounded set we can choose the dth-component of the function h to be $h_d(\theta, z_t) = \chi(\theta, z : h(\theta, z_t) \in A)$ where χ is the indicator function, i.e. $\chi(h(\theta, z_t) \ge A) = 1$ if $h(\theta, z_t) \in A$ and zero otherwise. Similarly, for any given α or H, we can compute $P[h(\theta, z_t) \le H] = \alpha$ by appropriately selecting the indicator function. Once quantiles and probability statements are available, we can evaluate whether the model can, in a probabilistic sense, reproduce features of the actual data.

Suppose, we have a vector of statistics H from the actual data and we are interested in the probability that H could be generated by the chosen parametrization of the model. One way to evaluate the model is to take the actual realization of the statistics as a critical value and compute the probability that $h(\theta, z_t)$ is less than or equal to H, i.e. evaluate the model's likelihood of realizing the vector of statistics we observe in the data⁷.

Another way to evaluate the model is to choose an α and, using the quantiles of the simulated \tilde{H} distribution, compute a critical value \tilde{H} satisfying $P[h(\theta, z_t) \leq \tilde{H}] \leq \alpha$. Comparing H and \tilde{H} would then give a one-sided procedure to evaluate the hypothesis that H has been generated by the model at a α % level.

4.2 Sensitivity Analysis

When one employs a Monte Carlo approach to compute integrals like (8) an automatic global sensitivity analysis on the support of the parameter space is performed as a by-product of the

⁷Alternatively, one can choose the set A to be the point estimate for the vector of statistics plus or minus one or two standard deviations, and then calculate the probability that the model generates functions $h(\theta, z_t)$ in the chosen set. Since for some of the statistics employed in this paper standard errors are unavailable, we do not report probability statements of this type.

simulations. Sensitivity analyses can, however, take other more specific forms. For example, one might be interested in evaluating the probability of an x_t path associated with a specific estimate of θ (say, e.g. the simulated method of moments estimator of θ) or, perhaps, in assessing what is the maximal variation in x_t or $h(\theta, z_t)$ which is consistent, say, with θ being within a two standard error band of a particular estimated value. To perform this type of analysis simply slice the joint density for θ and z_t in the appropriate dimensions, draw a time path for z_t and construct paths for x_t for one or more draws of θ in the particular range.

The approach to model evaluation we propose shares features with the procedure proposed by Gregory and Smith (1991). In their framework, however, parameters are calibrated. Since no allowance is made for parameter uncertainty, sensitivity analysis is roughly performed by replicating the experiment for different calibrated values of the parameters. Our approach also shares features with Kwan (1990). Similar to us he allows for parameter uncertainty in his simulation scheme but performs model evaluation by calculating the pairwise posterior odds-ratio for alternative model specifications. In other words, while we evaluate the model in an absolute sense, Kwan's procedure generates probability statements relative to other possible specifications ⁸.

4.3 Selecting $\pi(\theta|\mathcal{F})$

The selection of $\pi(\theta|\mathcal{F})$ is a crucial ingredient in our simulation procedure. One could choose it to be the asymptotic distribution of the SMM estimator of θ as in Canova and Marrinan (1990) or of the GMM estimator of θ as in Burnside, Eichenbaum and Rebelo (1990). Alternatively, one could chose it to be a "subjective" Bayesian prior as in Kwan (1990) or an "objective" one, as in Phillips (1991). Here we select $\pi(\theta|\mathcal{F})$ to reflect the cross study variation in existing econometric evidence and to be consistent with standard simulation practices. To be as uncontroversial as possible we choose $\pi(\theta|\mathcal{F})$ to be the frequency distribution of estimates of θ available in the literature, weighting estimates from all studies we are aware of equally ⁹. If no econometric evidence is available and economic theory does not provide a range for a parameter, we assume a uniform density on a support chosen on the basis of our own calculations. In addition, since existing information about

⁸Other approaches which use different criteria to evaluate simulated models have been proposed by Watson (1990) and King and Watson (1991). A different methodology to undertake sensitivity analysis has been suggested by Canova, Finn and Pagan (1991)

⁹We neglect the fact that since some studies use the same data sets, estimates for certain parameters are not independent. As long as the resulting estimates reflect sampling variability due to different estimation techniques or different sample sizes, dependence of the estimates does not create a problem here.

The parameters of the model can be divided into two groups: one includes those which have an economic interpretation (β, γ) and for which a rich set of estimates exists in the literature. We use this empirical evidence to construct frequency distributions of estimates in these dimensions. A second group includes all remaining parameters characterizing the distribution of the exogenous processes. For this second group the econometric evidence is scant or nonexistent and we express our ignorance by choosing a reasonable range for the support and imposing uniform densities in these dimensions.

For monthly data the discount factor β is typically estimated to be in the neighborhood of 0.996 with a small standard error (see e.g. Hansen and Singleton (1983) or Eichenbaum, Hansen and Singleton (1988)). The estimates vary from a minimum of 0.990 (see e.g. Hansen and Singleton) to a maximum of 1.0022 (see e.g. Dunn and Singleton (1986)). In general, estimates of β are not independent of estimates of the risk aversion parameter γ . For the studies we analyzed the rank correlation coefficient between estimates of γ and β is 0.12. When γ and β are jointly estimated, estimates of γ range between 0.5-1.5 when consumption of nondurable and services are used (see e.g. Hansen and Singleton (1983), Brown and Gibbons (1985), Mark (1985) or Heaton (1991)) to 2.5-3.5 when consumption of both durables and nondurables are used (see Dunn and Singleton (1986)) ¹⁰. In a study where the discount factor did not appear, Canova and Marrinan (1990) found that a value of γ close to zero best fit the data. In other studies where the discount factor (see is fixed the estimated value of γ is larger (see e.g. Burnside, Eichenbaum and Rebelo (1990)).

In simulation studies, β is typically chosen to produce a steady state real risk free rate of 1-5% on an annual basis (see e.g. Merha and Prescott (1985), Weil (1989), Giovannini and Labadie (1989), or Backus, Gregory and Zin (1989)). This implies that on a monthly basis a reasonable range for β is [0.9951, 0.9992]. On the other hand, the range for γ is much larger and varies from 0.5 to 55 (see e.g. Cooley and Ohanian (1990), Giovannini and Labadie (1989), Labadie (1989), Donaldson, Merha and Johnsen (1990), Backus, Gregory and Zin (1989) and Kandel and Stambaugh (1990)).

We capture this information by choosing the marginal density for the β to be truncated normal centered around 0.997, with range [0.990, 1.0022] and the marginal density for γ to be $\chi^2(4)$ with

¹⁰Many studies, following Friend and Blume (1970), estimated γ to be about 2. Kocherlakota (1990) shows that because of small sample biases estimates for γ of the order of 2 are consistent with a "true" value of about 13 (see also Kandel and Stambaugh (1990).

range [0, 55]. Since the rank correlation between estimates of γ and β is quite low, we assume that the joint density of these two parameters is the product of the two marginals ¹¹.

A few features of the two densities should be noted. The density of β is skewed to the left to conform to the idea that an annual real rate of 2-3% is more likely than a value in excess of 5%. The density for γ has mode at 2, which is the value most typically found in micro econometric studies and often used for benchmark simulations. In addition, it puts very low weights on high values of γ . The 95% range of a $\chi^2(4)$ is, in fact, [0.7, 10] and less than 1.0% of the mass of the density is in the region where γ exceeds 13.

The next 10 parameters $(A_{01}, A_{11}, a_{01}, a_{11}, a_{21}, A_{02}, A_{12}, a_{02}, a_{12}, a_{22})$ describe the conditional means and variances of output growth and money supply growth. Several studies document that the processes for output and the monetary base in the US appear to contain at least one unit root (see e.g. Stock and Watson (1989)). Using Citibase tape data we computed the first order autocorrelations for the growth rate of industrial production and of the monetary base to be respectively, .53 and .01 with standard errors equal to .07¹². We use this information by selecting a density for A_{11} to be uniform on [0.46, 0.60] and for A_{12} to have 50% of the mass uniformly distributed in the interval [-0.06, -0.00001] and 50% of the mass at 0. This implies that we give a fifty-fifty chance to the unit root hypothesis for the process (see Sims (1988) for the rationale for this representation). When output and the base have a unit root, A_{01} , A_{02} represent the average drift of the processes. Output in the US for the period 1964-1987 grew at an average rate of 0.2% per month with a standard deviation of 0.9%. The average growth rate of the base in the US has been 0.6% per month with a standard error of 0.3%. Therefore, we take A_{01} , A_{02} to be uniformly distributed over the intervals [-0.007, 0.011] and [0.003, 0.009].

Little information about the parameters of the variances of output and the base is available. Hodrick (1989) and Canova and Marrinan (1990) estimate the conditional variances of these processes using GARCH specifications. We incorporate the information contained in these two studies by selecting a uniform prior for all parameters: a_{11} and a_{12} have densities with support on [-0.002, 0.002]; a_{21} has support on [0.14, 0.38], and a_{23} on [0.06, 0.36] ¹³. Finally, we select a_{01} and a_{02}

¹¹Experiments conducted using a joint density for β and γ with a correlation of 0.12 did not change the results. ¹²Because of the high powered nature of the money supply in this model, we use the monetary base as opposed to

broader measures of monetary aggregates.

¹³Our point estimates of the GARCH coefficients for the three processes are the median values of the assumed bands. As an alternative, and since estimates of these parameters are conditionally normal, one could also draw from a joint normal distribution. We prefer uniform distribution because the GARCH parameters for these processes are

Ð

so that, given the values for a_{11} , a_{12} , a_{21} , a_{22} , the unconditional variances of the two simulated processes lies within a one standard error band around the point estimate of the unconditional variance over the 64-88 sample ([0.000001, 0.00001] and [0.00008, 0.0001] respectively).

For the remaining 5 parameters characterizing the behavior of government expenditure (A_{03} , A_{13} , a_{03} , a_{13} , a_{23}) no econometric evidence exists because data on the size of government expenditure shares in total output is not available at a monthly frequency. We collect an estimate of the unconditional mean and variance for government expenditure share in total output at a quarterly frequency. We find that this mean share is stable across time at around 0.04 and its variability is about 0.08. We impose these restrictions on our simulated share of government expenditure by choosing A_{03} to be uniform on [0.02, 0.06] and a_{03} to be uniform on [0.05, 0.10]. Since there is no information for setting (A_{13}, a_{13}, a_{23}) we chose them to be uniform in [0,1] but eliminate any draw which induces a time path for government expenditure share that, once it is aggregated at quarterly frequency, is inconsistent with the reported quarterly evidence.

The final ingredient required is the choice of initial conditions for the exogenous processes. To make the simulations comparable with the actual data we chose as initial conditions the realized values for the exogenous processes in 1964,6.

The Results 5

Tables 4-6 report statistics for yields at 1-3-6-12 month maturities, for six spreads and for three cross correlations when 10000 simulations were performed ¹⁴. For each statistic we report a simulated 90% band and the probability that the model generates a value less than or equal to the value observed in the data.

The tables indicate that the model reproduces several qualitative features of the US term structure: the average yield curve slopes upward (the probability that the term structure is upward sloping is 0.97), the volatility of yields decreases with maturity, yields of 3-6-12 month maturities exhibit a high degree of serial correlation and changes in short term yields are positively correlated with changes in long term yields.

rather imprecisely estimated.

¹⁴All the simulation reported are performed on a VAX 8700 machine using RATS programs, which are available on request from the authors. Codes drawing random numbers from the chosen densities are also available on request. We produced our own pseudorandom numbers because we found that the periodicity of the algorithms creating random numbers in standard statistical software is too short to avoid repetitions. Our algorithm, which is based on Press, et. al. (1989), passes the 12 tests for randomness of Knuth (1981) and has a periodicity of 714025.

Quantitatively, the model matches the variability of yields and the higher moments but has three shortcomings. First, the 90% bands for the mean of yields is slightly too low. Contrary to Backus, Gregory and Zin (1989), a high variability of yields is obtained here only at the cost of producing very low (or even negative) values for their mean. Second, the bands for the statistics testing for heteroskedasticity are also slightly too low. Third, and more importantly, the model fails to produce enough serial correlation in the simulated data. On average, simulated yields have first order serial correlation coefficients which are 25% lower than what we observe in the US data.

The model is relatively more successful in accounting for the quantitative properties of the spreads. Only the mean of the spreads at the lowest end of the term structure and the level of heteroskedasticity are at odds with the actual data. While the former failure is significant, the latter one is minor. Because the model generates, approximately, the same amount of heteroskedasticity in all the yields, the spreads fail to be as heteroskedastic as we observe in the actual data.

Finally, the simulated correlations between changes in long term yields and changes in short term yields and correlations between forward premia and future changes in yields and between spreads and future changes in yields are too high to be consistent with the data.

To intuitively understand how the model can reproduce important qualitative features of the US term structure previously unexplained, consider its simplest version where there is no government and no money ($G_t = M_t = 0$).

The pricing formula at t for an asset delivering one unit of the consumption good at t + k with certainty is:

$$V_{t,k} = \beta^k E_t \frac{U'(C_{t+k})}{U'(C_t)} = \beta^k E_t (\frac{Y_{t+k}}{Y_t})^{-\gamma}$$
(10)

The yield on this asset at time t is

$$r_{t,k} = -\frac{\ln V_{t,k}}{k}$$

$$\approx -\ln\beta + \frac{\gamma}{k} E_t (\ln[\frac{Y_{t+k}}{Y_t}]) - \frac{\gamma^2}{2k} var_t (\ln[\frac{Y_{t+k}}{Y_t}])$$
(11)

where $E_t(.)$ and $var_t(.)$ refer to the conditional mean and variance of the quantity in parenthesis and the approximation comes from the truncation in the Taylor expansion.

From (11) it is clear that time variation in the conditional variance of the exogenous processes of the economy is a potentially important determinant of the cyclical behavior of interest rates. For example, if the process for output is a random walk, time variation in the variance of output growth entirely accounts for variation over time in yields. In addition, when future economic uncertainty is large, a riskless bill is more highly valued and, consequently, its yield may be very low (even negative). The very low average value of the risk free rate observed in the US has been considered by many troublesome (see e.g. Weil (1989)). High variability in the exogenous processes may account for this behavior (see Huggett (1991) for an alternative explanation). Note also that as $k \to \infty$, $var_t(\ln[\frac{Y_{t+k}}{Y_t}]) \to var(\ln[\frac{Y_{t+k}}{Y_t}])$ unless the conditional variance of output growth is very persistent. Therefore, variation in the uncertainty surrounding the driving processes far in the future will have no effect on current yields. This implies that heteroskedasticity in the exogenous forces of the economy is likely to impact primarily on the shorter end of the term structure.

The unconditional autocovariance function of yields (and spreads) also depends on the presence of conditional heteroskedasticity in a nontrivial way. For example, the unconditional variance of the yield on a bill of maturity k is:

$$var(r_{t,k}) = E[\frac{\gamma}{k}(E_t(\ln[\frac{Y_{t+k}}{Y_t}]) - E(\ln[\frac{Y_{t+k}}{Y_t}])) - \frac{\gamma^2}{2k}(var_t(\ln[\frac{Y_{t+k}}{Y_t}]) - var(\ln[\frac{Y_{t+k}}{Y_t}]))]^2$$

$$= \frac{\gamma^2}{k^2}E[\sum_{j=0}^{k-l}A_1^j\ln[\frac{Y_t}{Y_{t-1}}] + A_0\sum_{l=1}^k\sum_{j=0}^{l-1}A_1^j - \frac{kA_0}{1 - A_1}]^2$$

$$+ \frac{\gamma^4}{4k^2}E[\sum_{l=1}^k\sum_{j=0}^{k-l}A_1^{2j}(\sigma_{t+l}^2 - E(\sigma_{t+l}^2))]^2$$
(12)

If output is a random walk, the first term in (12) drops out and, if there was no heteroskedasticity, the variance of interest rates would be identically equal to zero. When conditional heteroschedasticity is present, the variance of interest rates depends on the signs and relative magnitudes of the GARCH parameters, the maturity of the bill and the size of the deviations of the conditional from unconditional variability of output. Similarly, using the fact that the autocorrelation function of yields can be computed as $corr(r_{t,k}, r_{t-k,k}) = \frac{E(1+R_{t,2k})-[E(1+R_{t,k})]^2}{var(r_{t,k})}$ (see e.g. Kandel and Stambaugh (1990)), is immediate to note that heteroskedasticity in output growth will have an impact on the entire second order properties of yields. Therefore, time variation in the conditional second moments of the driving processes may be crucial in matching the variability and the correlation structure of yields at the short end of the term structure, especially when the driving processes are nearly integrated.

To confirm the intuition provided with the above simple analytical example, we conduct a numerical experiment using conditionally homoschedastic exogenous processes. Tables 7-9 present

the results and display several interesting features. First, the standard errors of yields are very small and the bands are narrow. Second, the third and fourth moments of yields are much smaller in absolute value than in the heteroskedastic case, the bands are shifted toward positive values and the median of the band is always around zero. Third, the behavior of the bands for the bands are smaller in size and shifted toward zero. For 12 month yields, the upper tail of the distribution almost completely disappears. The behavior of the spreads tracks very closely the behavior of yields. Finally, with homoschedastic exogenous processes cross correlations are very different from the heteroskedastic case. The median value of the contemporaneous cross correlation of changes in short and in long term yields drops significantly and the lower tail of the distribution includes, in two cases, negative values. The bands for the cross correlations of both forward premia and spreads with future changes in yields move toward zero. As expected from the above discussion, the bands for the cross correlations of longer term forward premia and spreads with future changes in yields are the least affected by the change.

In conclusion, the presence of heteroskedasticity in the exogenous processes appears to be important in reproducing the conditional and the unconditional variance of yields. It also helps in boosting the autocorrelation function of simulated data toward that of the actual data but does not quite do the job. The cost of introducing heteroskedastic processes in the model materializes primarily in higher values for higher moments of the yields and in the extreme values for the cross correlations between changes in short with long term yields and between forward premia and spreads with future changes in yields.

6 Some Sensitivity Results

Some of the assumptions we made in either solving the model or in specifying the nature of the stochastic processes may be considered controversial. In this section we examine the robustness of the conclusions obtained to modifications of these assumptions. We also examine whether it is the uncertainty present in the economic parameters or in the parameters characterizing the exogenous processes which is responsible is for the large size of the bands appearing in tables 4-6.

In deriving (2)-(5) we imposed the quantity theory. Hodrick, Kocherlakota, Lucas (1991) show that when a version of the above model is calibrated to the US economy the cash-in-advance con-

Û

straints almost always bind. Therefore, there appeared to be little practical gain in specifying models with more complicated nonbinding constraints. However, in principle, we can abstract from this problem entirely by simply taking the stochastic processes for consumption and prices as the primitives for our simulations. In practice, the quality of consumption data is poor. Wilcox (1988) pointed out that monthly aggregate consumption data is primarily interpolated from observations obtained at a much lower frequency. This interpolation procedure generates serially correlated measurement errors and disturbing autocorrelation properties in the data ¹⁵. In addition Breeden, Gibbons and Litzenberger (1989) indicated that summation biases may make the statistical properties of quarterly consumption data dubious as well.

With these caveats in mind we use (1) as our asset pricing equation. To perform simulations we must to select 10 new parameters. The densities for the parameters of consumption growth and inflation processes are uniform centered around the point estimates with ranges equal to a one standard error band. Estimates of these parameters are obtained using the monthly consumption data on nondurables and services and the monthly personal consumption expenditure index 16. Letting $z_t = [\Delta \log P_t, \Delta \log C_t]$, the ranges for the ten parameters characterizing the two processes are: $A_{01} \in [0.0018, 0.0022], A_{11} \in [0.30, 0.38], a_{01} \in [0.0000026, 0.0000034], a_{11} \in [-0.0001, 0.0001],$ $a_{21} \varepsilon$ [0.20, 0.44], $A_{02} \varepsilon$ [0.0017, 0.0023], $A_{12} \varepsilon$ [-0.32, -0.22], $a_{02} \varepsilon$ [0.000009, 0.000011], $a_{12} \varepsilon$ [-0.0008, 0.0006], $a_{22} \varepsilon$ [0.02, 0.07].

Tables with the outcomes of this and other experiments are reported in an appendix available from the authors. Here we briefly summarize the main features of the results. We find that the probability that the model generates an upward sloping yield curve drops to about 55%. Backus, Gregory and Zin (1989) demonstrated that, on average, an upward sloping yield curve obtains when the growth rates of the driving processes are negatively serially correlated. In the present instance, the first order serial correlations of consumption growth and inflation are approximately of the same order but of opposite signs. Therefore, in large samples, one should expect that approximately in 50% of the simulations the average term structure will slope upward.

We also find that the variability and the amount of serial correlation and of heteroskedasticity in simulated yields and spreads decreases. However, qualitatively, none of the features reported in

¹⁵The first order serial correlation coefficient of monthly consumption growth for the data set we use is -0.27. If we aggregate the monthly data at a quarterly level the correlation becomes 0.13. For published quarterly data on consumption growth the first order correlation coefficient is 0.29.

¹⁶This data was kindly provided by Masao Ogaki.

There is some evidence in the literature (see e.g. Spanos (1990)) that GARCH models fail to capture important distributional characteristics of many processes. We conducted experiments using alternative functional forms for the conditional variances (as in Schwert (1990)) or using nonparametric estimates of the conditional moments (as in Pagan and Ullah (1988)). We found numerical changes in the reported statistics but the essence of the results is unaltered.

Lewis (1991) presents evidence that the unconditional distribution of one component of z_t is nonstationary. She argues that the uncertainty due to regime changes in monetary policy in the US may have had a non-negligible impact on the behavior of the term structure for the 79-82 period. In addition to this, when a process is subject to structural shifts, estimates of the conditional variance obtained from GARCH (AR or nonparameteric) models are biased, tend to understate the true conditional variance of the processes (see Evans (1991)) and may affect the time series properties of simulated yields.

To examine the impact of a change in the unconditional distribution of the monetary base on the term structure of yields, we performed simulations drawing the parameters of the process for the base from two different densities: one which matches the properties of the base before 1979 (first subsample consisting of time periods 1 through 176) and a second one that matches its properties for the period 1979-87 (second subsample consisting of time periods 177 though 270) ¹⁷.

Estimates of the AR-GARCH parameters of the growth rate of the base are approximately identical over the two subsamples, except for the first order serial correlation coefficient. Before 1979, A_{11} is estimated to be -0.40 with a standard error of 0.14. After 1979, A_{11} is estimated to be 0.31 with the same standard error (compare with a value of -0.03 and a standard error of 0.07 obtained for the entire sample). The densities for the five parameters characterizing the conditional moments of money growth rates are assumed to be uniform centered around the point estimate of the parameter with the following ranges: before 1979 $A_{03} \varepsilon$ [0.010, 0.012], $A_{13} \varepsilon$ [-0.54, -0.26], $a_{03} \varepsilon$ [0.000004, 0.000006], $a_{13} \varepsilon$ [-0.0001, 0.0001], $a_{23} \varepsilon$ [0.01, 0.35] and after 1979 $A_{03} \varepsilon$ [0.003, 0.005], $A_{13} \varepsilon$ [0.17, 0.45], $a_{03} \varepsilon$ [0.000009, 0.00012], $a_{13} \varepsilon$ [-0.00008, 0.00008], $a_{23} \varepsilon$ [0.01, 0.38].

¹⁷A more appropriate way to check whether Lewis's objection is really relevant would be to estimate the parameters of the base recursively (with the Kalman filter) and draw parameters in the simulation from recursive densities (one for each of the 270 time periods generated in each simulation). Because of the complexity of the operation and of the limited computer capabilities available to us we did not perform this exercise.

The results indicate that this modification improves the performance of the model. Both the serial correlation and the heteroskedasticity present in simulated yields and spreads increases. The model now generates about 90% of the serial correlation we observe in actual yields and can account for their heteroskedastic structure. In addition, the contemporaneous cross correlations of changes in yields are much lower than in the basic case. In three out of the five cases, the correlations observed in the data can be generated by the model with reasonable probability ¹⁸.

To determine whether is it the uncertainty in the estimates of the parameters of the exogenous processes or the uncertainty we face in chosing values for the "economic" parameters which is responsible for the size of bands reported in the tables, we conduct two additional experiments. Each experiment involves slicing the joint distribution of parameters and exogenous processes in different dimensions.

In the first case, we "calibrate" the stochastic process for the exogenous variables by selecting point estimates for the parameters of their conditional means and variances (which are the midpoints of the ranges described in section 4.3). This experiment should indicate how the uncertainty surrounding the parameters of the stochastic processes is reflected in the bands for the reported statistics. In the second case, we "calibrate" the two economic parameters $(\beta = .997, \gamma = 0.0, 2.0, 10.0)$ and examine the distributions of the statistics when the parameters of the exogenous processes are randomly drawn. This experiment indicates how sensitive the results are to uncertainty in the economic parameters economists care about most.

We find that the simulated statistics are somewhat sensitive to the uncertainty in the parameters of the exogenous processes. When we fix these parameters the 90% bands for the moments of all yields are tighter, the band for the standard errors are lower and, on average, there is less serial correlation and much less heteroskedasticity in the simulated processes. Similar results emerge for the spreads.

When β and γ are fixed at some "reasonable" value we again find that the bands for the reported statistics change substantially. In particular, the absolute level of the standard error of simulated yields and spreads drops substantially. As we increase γ from 0 up to 10, the whole band for the

¹⁸We also introduced international considerations in the model to gauge whether the addition of new potential sources of time variation in the simulations (variations in the second moments of foreign output and money) helped to quantitatively improve our fit of the cyclical properties of yields. We found that the impact of international factors on the term structure is small but that both serial correlation and heteroskedasticity decrease. This is because both the persistence and the deviations of the conditional from the unconditional variance of foreign variables are much smaller than those of US variables.

autocorrelations is shifted toward zero for the one month rate and change nonmotonically for the 6 and 12 month rates. This change is achieved at the cost of shifting the band for mean yields and introducing a large amount of skewness and kurtosis in the simulated data. As in Cooley and Ohanian (1990), we find that values of γ close to zero minimize the distortions in the second order properties of the simulated data.

7 Conclusions

This paper attempted to reconcile the US term structure of interest rates with the predictions of a standard monetary consumption based ICAP model. We modified the basic model to allow for conditional heteroskedasticity in the exogenous processes of the economy and found the modification helpful in accounting for some puzzling features of the yield curve.

We show that the model can reproduce the average slope of the yield curve, the absolute variability of yields and the fact that volatility decreases with maturity and comes close to (but falls short of) matching the serial correlation properties of yields. The model is more successful in accounting for features of the spreads at various maturities. For almost all statistics examined the actual statistic observed in US data falls within the 90% bands and, in a large number of cases, the actual statistics fall near the medians of the simulated distributions. The model produces with high probability contemporaneous cross correlations for changes in long and short term yields which are, in general, too large and cross correlations at leads which are too small to be consistent with US evidence. The same is true for forward premia and future changes in yields. When a break in the unconditional distribution of the monetary base is allowed, this shortcoming is partially eliminated. This is not the case for correlations between the spreads and future changes in yields. This failure is important and deserves further study.

Although the representative agent paradigm is ill-suited to understand the complexity of financial markets, we believe that further experiments with this model are necessary to discover what features of the real world are consistent with the approach before proceeding to more complex multi-agent specifications (as e.g. in Marcet and Singleton (1991), Heaton and Lucas (1991) or Telmer (1991)). Extensions of this single agent model to include capital and variable labor as in DenHaan (1991) or to consider the production-based ICAP model of Cochrane (1991) or some form of liquidity constraint as in Lucas (1991) or Huggett (1991) are likely to be fruitful in eliminating some of the problems reported here. We do not believe, however, that the introduction of habit persistence, along the lines of Costantinides (1990), is the key to solve the problems we have highlighted. Habit persistence helps to increase the variability of yields at the cost of shifting the mean of the entire yield curve toward unreasonably low or negative values.

Two interesting aspect of the data we have ignored in this study are the behavior of the term structure at turning points and the relationship between the term structure and inflation. In the actual data, there appears to be an inversion of the yield curve right before a turning point (see e.g. Stambaugh (1988)) and a tight relationship between the term structure and future inflation (see e.g. Mishkin (1988)). In a future study we plan to examine whether a version of the ICAP model with time varying conditional variances can reproduce these features of the actual data.

Appendix

The closed form solution for the interest rate on a bill of maturity k, is given $\forall k$ by:

$$\begin{aligned} r_{i,t}^{k} &= -\ln\beta + \frac{1}{k} (\mathcal{G}_{t}(k) + \gamma \log(1 - z_{3t})) \\ &+ \sum_{j=1}^{k} (A_{12}^{j} z_{2t} + A_{02} \sum_{l=1}^{k} \sum_{j=0}^{l-1} A_{12}^{j})] - 0.5 (\sum_{l=1}^{k} \sum_{j=0}^{l-1} A_{12}^{2j} (a_{12}^{l-j} \sigma_{2t}^{2} + a_{02} \sum_{m=0}^{l-j-1} a_{12}^{m} + a_{12}^{l-j-1} a_{22} \epsilon_{2t}^{2})] \\ &+ (\gamma - 1) [(\sum_{j=1}^{k} (A_{11}^{j} z_{1t} + A_{01} \sum_{l=1}^{k} \sum_{j=0}^{l-1} A_{11}^{j})] \\ &- 0.5 * [(\gamma - 1)]^{2} * (\sum_{l=1}^{k} \sum_{j=0}^{l-1} A_{11}^{2j} [a_{11}^{l-j} \sigma_{1t}^{2} + a_{01} \sum_{m=0}^{l-j-1} a_{11}^{m} + a_{11}^{l-j-1} a_{21} \epsilon_{1t}^{2}] + u_{i,t}^{k}) \end{aligned}$$
(13)

where $u_{i,t}^k$ is the approximation error and where $\mathcal{G}_t(k)$ involves the parameters of the process for government expenditure share and γ and is given by:

$$\mathcal{G}_{1t}(k) = -\ln[-(1 - A_{13}^k z_{3t} - A_{03} \sum_{j=0}^{k-1} A_{13}^j - \sqrt{12\mathcal{H}_3(k)})^{(1-\gamma)} + (1 - A_{13}^k z_{3t} - A_{03} \sum_{j=1}^{k-1} A_{13}^j)^{(1-\gamma)}] - \ln[\sqrt{12\mathcal{H}_3(k)}] - \ln(1-\gamma)$$
(14)

where $\mathcal{H}_3(p) = (a_{13}^p \sigma_{3t}^2 + a_{03} \sum_{j=0}^{p-1} a_{13}^j + a_{23} a_{13}^{p-1} \epsilon_{3t}^2).$

The forward rate, the spread and the forward premium between interest rates of maturity k and h can then be computed directly from (14).

References

- Backus, D., A. Gregory and S. Zin, 1989, "Risk Premiums in the Term Structure: Evidence from Artificial Economies", *Journal of Monetary Economics*, 24: 371-399.
- Braun, T., 1990, "The Interaction of Distortionary Taxes and Aggregate Variables in PostWar Data", University of Virginia, manuscript.
- [3] Breeden, D., 1979, "An Intertemporal Asset Pricing Model with Stochastic Consumption and Investment Opportunities", Journal of Financial Economics, 7: 265-296.
- [4] Breeden, D., 1986, "Consumption, Production, Inflation and Interest Rates: A Synthesis", Journal of Financial Economics, 16: 3-39.
- [5] Breeden, D., Gibbons, M. and Litzenberger, 1989, "Empirical Tests of the Consumption oriented CAPM", Journal of Finance, 44: 231-262.
- [6] Breush T.S. and A. Pagan, 1979, "A Simple Tests for Heteroschedasticity and for Random Coefficient Variation", *Econometrica*, 49: 1287-1294.
- [7] Brown, S. and Gibbons, M., 1985, "A Simple Econometric Approach for Utility Based Asset pricing Models", *Journal of Finance*, 40: 338-359.
- [8] Burnside, C., Eichenbaum, M. and Rebelo, S., 1990, "Labor Hoarding and the Business Cycle", NBER working paper 3556.
- [9] Campbell, J., 1986, "A Defence of the Traditional Hypothesis about the Term Structure of Interest Rates", Journal of Finance, 41: 183-193.
- [10] Campbell, J., 1987, "Stock Returns and the Term Structure", Journal of Financial Economics, 17: 373-399.
- [11] Campbell, J. and Shiller, R., 1984, "A Simple Account of the Behavior of the Long Terms Interest Rates", American Economic Review, Papers and Proceedings, 74: 44-48.
- [12] Campbell, J. and Shiller, R., 1987, "Cointegration and Tests of Present Value Models", Journal of Political Economy, 95: 1062-1088.

- [13] Campbell, J. and Shiller, R., 1990, "Yield Spread and Interest Rates Movements: A Bird's Eye View", NBER working paper 3153
- [14] Canova, F., 1990, "Sensitivity Analysis and Model Evaluation in Simulated Dynamic General Equilibrium Economies", University of Rochester, manuscript.
- [15] Canova, F. and J. Marrinan, 1990, "Profits, Risk and Uncertainty in Foreign Exchange Markets", Brown University, working paper 8935.
- [16] Cochrane, J., 1991, "Using Production Based Asset Prices to Explain the Behavior of Stock Returns over the Business Cycle" *Journal of Finance*, 46: 209-238.
- [17] Cooley, T and Ohanian, L., 1990, "Yield Curve Inversion and Real Activity', University of Rochester, manuscript.
- [18] Costantinides, G., 1990, "Habit Persistence: A Resolution of the Equity Premium Puzzle" Journal of Political Economy, 98: 519-543.
- [19] DenHaan, W., 1990, "Money, Leisure and the Term Structure of Interest Rates", Carnegie Mellon University, manuscript.
- [20] Diebold, F., 1988, Empirical Modelling of Exchange Rate Dynamics, (Springer Verlag, New York, N.Y.).
- [21] Donaldson, J., T, Johnsen and R. Merha, 1990, "The Behavior of the Term Structure of Interest Rates in a Real Business Cycle Model", *Journal of Economic Dynamic and Control*, 14: 571-598.
- [22] Duffie, D. and Singleton, K., 1990, "Simulated Moment Estimator of Markov Models of Asset Prices", NBER Technical Working Paper 87.
- [23] Dunn, K. and Singleton, K., 1986, "Modelling the Term Structure of Interest Rates under Nonseparable Utility and Durability of Goods", *Journal of Financial Economics*, 17: 27-55.
- [24] Eichenbaum, M, L. Hansen and K. Singleton, 1988, "A Time Series Analysis of Representative Agent Models of Consumption and Leisure Choice under Uncertainty", *Quarterly Journal of Economics*, 103: 51-78.

- [25] Engle, R., D. Lilien and R. Robins, 1987, "Estimating Time Varying Risk Premia in the Term Structure: the ARCH-M model", *Econometrica*, 55: 391-405.
- [26] Evans, M., 1991, ".", Journal of Money, Credit and Banking
- [27] Evans, M. and Lewis, K., 1990, "Do Risk Premia Explain it All? Evidence from the Term Structure", New York University, manuscript.
- [28] Fama, E., 1976, Foundation of Finance, (Basic Books, New York, N.Y.).
- [29] Fama, E., 1984, "The Information in the Term Structure", Journal of Financial Economics, 13: 509-528.
- [30] Ferson, W. and Harvey, C., 1991, "The Variation in Economic Risk", Journal of Political Economy, 99: 385-415.
- [31] Friend, I. and Blume, M., 1970, "Measurement of Portfolio Performance Under Uncertainty", American Economic Review, 60: 561-576.
- [32] Froot, K., 1989, "A New Hope for the Expectation Hypothesis of the Term Structure of Interest Rates", Journal of Finance, 44: 283-305.
- [33] Giovannini, A. and Labadie, P., 1989, "Asset Prices and Interest Rates in Cash-in-Advance Models", Institute for International Economic Studies, University of Stockholm, working paper.
- [34] Gelfand, A. and A. Smith, 1990, "Sampling Based Approaches to Calculating Marginal Densities", Journal of the American Statistical Association, 85: 398-409.
- [35] Geweke, J., 1989, "Bayesian Inference in Econometric Models using Monte Carlo Integration", Econometrica, 57: 1317-1339.
- [36] Hansen, L. and Hodrick, R., 1983, "Risk Averse Speculation in Forward Foreign Exchange Markets: An Econometric Analysis of Linear Models" in J. Frenkel, ed., *Exchange Rates and International Macroeconomics*, (Chicago, University Press, Chicago, II.).
- [37] Hansen, L. and Singleton, K., 1983, "Stochastic Consumption, Risk Aversion and the Temporal Behavior of Asset Returns", *Journal of Political Economy*, 91: 249-265.

- [38] Heaton, J., 1991, "An Empirical Investigation of Asset Pricing with Temporally Dependent Preferences Specifications", MIT, manuscript.
- [39] Heaton, J. and Lucas, D. 1991, "Asset Pricing, Undiversifiable Income Shocks and Trading Costs, MIT, manuscript.
- [40] Hugget, M., 1991, "The Risk Free Rate in Heterogeneous Agent, Incomplete Insurance Economies", University of Illinois, manuscript.
- [41] Hodrick, R., Kocherlakota N. and Lucas, D., 1991, "The Variability of Velocity in Cash-in-Advance Models", Journal of Political Economy, 99, 358-384.
- [42] Kandel, S. and Stambaugh, R., 1990, "Expectations and the Volatility of Consumption and Asset Returns", *Review of Financial Studies*, 3: 207-232.
- [43] Kendall M. and Stuart, A., 1958, The Advanced Theory of Statistics, vol.2, (Hafner, New York, N.Y.).
- [44] Kessel, R., 1965, "The Cyclical Behavior of the Term Structure of Interest Rates", NBER occasional paper 91.
- [45] King, R. and Watson, M., 1991, "Comparing the Fit of Alternative Dynamic Models", UNiversity of Rochester, manuscript.
- [46] Knuth, D., 1981, Seminumerical Algorithms, Second Edition, Volume 2 of The Art of Scientific Computing, (Addison and Wesley, Reading, Ma.).
- [47] Kocherlakota, N., 1990, "On Tests of Representative Consumer Asset Pricing Models", Journal of Monetary Economics, 26: 285-304.
- [48] Kwan, Y.K, 1990, "Bayesian Model Calibration with an Application to a Non-linear Rational Expectation Two-Country Model", University of Chicago, Business School, manuscript.
- [49] Labadie, P., 1989, "Stochastic Inflation and the Equity Premium", Journal of Monetary Economics, 24: 277-298.
- [50] Labadie, P. 1991, "The Term Structure of Interest Rates over the Business Cycle", Federal Reserve Board, manuscript.

- [51] Lee, B.S., 1989, "A Nonlinear Rational Expectations Models of the Term Structure of Interest Rates with Time Varying Risk Premia", *Journal of Money, Credit and Banking*, 21: 347-367.
- [52] Lee, B.S. and Ingram, B., 1991, "Simulation Estimators of Time Series Models", Journal of Econometrics, 47: 197-207.
- [53] Lewis, K., 1991, "Was there a 'Peso Problem' in the US Term Structure of Interest Rates: 1979-1982", International Economic Review, 32: 159-179.
- [54] Ljung, G. and Box, G., 1978, " On the Measure of the Lack of Fit in Time Series Models", Biometrika, 67: 297-303.
- [55] Longstaff, F. and Schwartz, E., 1990, "Interest Rate Volatility and the Term Structure: A Two Factor General Equilibrium Model", Anderson Graduate School of Management, UCLA, manuscript.
- [56] Lucas, R., 1978, "Asset Prices in an Exchange Economy", Econometrica, 46: 1429-1445.
- [57] Lucas, R., 1980, "Equilibrium in a Pure Currency Economy", Economic Inquiry, 18: 203-220.
- [58] Mankiw, G., 1986, "The Term Structure of Interest Rates Revisited", Brookings Papers on Economic Activities, 6: 61-96.
- [59] Mankiw, G and Summers, L., 1984, "Do Long-Term Interest Rates Overreact to Short Term Interest Rates ?", Brookings Paper on Economic Activity, 1: 223-242.
- [60] Marcet, A. and Singleton, K. (1991) "Equilibrium Asset Prices and Savings of Heterogeneous Agents in the Presence of Portfolio Constraints", Stanford University, manuscript.
- [61] Mark, N., 1985, "On the Time Varying Risk Premia in Foreign Exchange Markets: an Econometric Analysis", Journal of Monetary Economics, 16: 3-18.
- [62] Merha, R. and Prescott, E., 1985, "The Equity Premium: A Puzzle", Journal of Monetary Economics, 15: 145-161.
- [63] Mishkin, F., 1988, "The Information in the Term Structure: Some Further Results", NBER working paper 2575.

- [64] McCulloch, H. and Shiller, R., 1987, "The Term Structure of Interest Rates", NBER working paper 2341.
- [65] Niedderreter, H., 1988, "Quasi Monte Carlo Methods for Multidimensional Numerical Integration" International Series of Numerical Mathematics, 85: 157-171.
- [66] Pagan, A. and Ullah, A., 1988, "The Econometric Analysis of Models with Risk Terms", Journal of Applied Econometrics, 3: 87-105.
- [67] Press, W, Flannery, B., Teukolsky, S. and Vetterling, W., 1989, Numerical Recipes: The Art of Scientific Computing, (Cambridge University Press, Cambridge, Ma.).
- [68] Phillips, P.C.B., 1991 "To Criticize the Critics", forthcoming, Journal of Applied Econometrics.
- [69] Rouwenhorst, G., 1990, "Asset Returns and the Business Cycle: A General Equilibrium Approach", University of Rochester, manuscript.
- [70] Rubin, D., 1987, Comment on "The Calculation of Posterior Distribution by Data Augmentation" by M. Tanner and W. Wong, *Journal of the American Statistical Association*, 82: 543-546.
- [71] Schotman, P., 1991, "When Unit Roots Matter:Excess Volatility and Excess Smoothness of long Term Interest Rates" Institute for Empirical Macroeconomics, Federal Reserve Bank of Minneapolis, working paper 44.
- [72] Schwert, W., 1990, "Stock Volatility and the Crash of 87", Review of Financial Studies, 3: 77-102.
- [73] Sims, C., 1988, "Bayesian Skepticism on Unit Root Econometrics", Journal of Economic Dynamic and Control, 12: 463-476.
- [74] Shiller, R., 1979, "The Volatility of Long Term Interest Rates and Expectations Theories of the Term Structure", Journal of Political Economy, 87: 1190-1219.
- [75] Shiller, R, Campbell, J and K. Schoemholtz, 1983, "Forward rates and Future Policy: Interpreting the Term Structure of Interest Rates", *Brooking Papers on Economic Activity*, 1: 173-219

- [76] Smith, T., 1990, "Estimation of Dynamic Economic Models by Simulation: A Comparison of the Approaches", Queen's University, manuscript.
- [77] Stambaugh, R., 1988, "The Information in Forward Rates: Implications for Models of the Term Structure", Journal of Financial Economics, 21: 41-70.
- [78] Startz, R., 1982, "Do Forecast Errors or Term Premia Really Make the Difference between Long and Short Rates?" Journal of Financial Economics, 10: 323-329.
- [79] Stock, J. and Watson, M., 1989, "Interpreting the Evidence of Money-Income Causality.", Journal of Econometrics, 40: 161-182.
- [80] Tanner, M. and W. Wong, 1987, "The Calculation of Posterior Distributions by Data Augmentation" (with discussion), Journal of the American Statistical Association, 82: 528-550.
- [81] Telmer, C., 1991, "Asset Pricing Puzzles and Incomplete Markets", Queen's University, manuscript.
- [82] Watson, M., 1990, "Measures of Fit for Calibrated Models" Federal Reserve of Chicago, working paper.
- [83] Weil, P., 1989, "The Equity Premium Puzzle and the Risk Free Rate", Journal of Monetary Economics, 24: 401-421.
- [84] Wilcox, D., 1988, "What Do We Know about Consumption ?", Board of Governors, Federal Reserve System, manuscript.
- [85] White, H., 1980, "A Heteroschedastic Consistent Covariance Matrix Estimator and Direct Tests for Heteroschedasticity', *Econometrica*, 48: 817-838.

	1 month T-Bill	3 months T-Bill	6 months T-Bill	12 months T-Bill
Mean	6.74	7.09	7.34	7.53
P-value	.000	.000	.000	.000
Std. Error	2.78	2.88	2.87	2.73
Skewness	-0.73	-0.78	-0.57	0.39
P-value	.000	.000	.000	.000
Kurtosis	8.86	7.48	7.81	3.26
P-value	.000	.000	.000	.000
Autocor. 1	0.948	0.965	0.966	0.974
Autocor. 2	0.904	0.922	0.924	0.944
Autocor. 4	0.823	0.826	0.857	0.879
Autocor.12	0.650	0.676	0.687	0.700
Autocor.24	0.398	0.392	0.414	0.443
ARCH(13)	37.20	43.17	40.32	47.22
P-value	.000	.000	.000	.000
BP(13)	75.44	56.91	56.19	54.80
P-value	.000	.000	.000	.000
White(26)	93.01	71.86	76.95	78.79
P-value	.000	.000	.000	.000
Q(49)	55.90	60.49	6.76	33.99
P-value	.231	.125	1.000	.949
KS	0.784	0.862	0.561	0.910

Table 1 Monthly Statistics: T-Bill Yields 1964,6-1987,11

Note: BP refers to Breusch-Pagan test, Q to the Ljung-Box test, KS refers to the Kolmogorov-Smirnov statistics. The numbers in parenthesis after ARCH, BP, White and Q refer to the number of degrees of freedom of the χ^2 statistics.

Table 2 Monthly Statistics:T-Bill Spreads 1964,6-1987,11

	1-3 months	1-6 months	1-12 months	3-6 months	3-12 months	6-12 months
	0.05	0.00	0.70	0.05	0.44	0.10
Mean	0.35	0.60	0.79	0.25	0.44	0.18
P-value	.000	.000	.000	.000	.036	.000
Std. Error	0.44	0.53	0.78	0.24	0.67	0.51
Skewness	1.23	0.84	0.21	0.37	0.33	-0.63
P-value	.000	.000	.039	.004	.008	.000
Kurtosis	8.13	5.11	7.33	2.42	15.39	24.38
P-value	.000	.000	.000	.000	.000	.000
Autocor. 1	0.221	0.338	0.491	0.587	0.672	0.611
Autocor. 2	0.198	0.330	0.309	0.449	0.421	0.296
Autocor. 4	0.039	0.068	-0.010	0.135	0.176	0.104
Autocor.12	0.361	0.209	0.173	0.098	0.154	0.131
Autocor.24	0.265	0.241	0.132	0.051	-0.063	-0.017
ARCH(13)	34.01	57.18	92.14	47.36	62.69	89.42
P-value	.000	.000	.000	.000	.000	.000
BP(13)	55.82	59.20	55.68	22.84	87.77	105.99
P-value	.000	.000	.000	.000	.000	.000
White(26)	75.26	128.57	139.86	72.24	162.01	141.62
P-value	.000	· .000	.000	.000	.000	.000
Q(49)	55.84	26.84	2.68	67.48	69.12	43.29
P-value	.233	.995	1.00	.041	.030	.702
KS	0.251	0.973	1.030	0.624	0.817	0.555

Note: BP refers to Breusch-Pagan test, Q to the Ljung-Box test , KS refers to the Kolmogorov-Smirnov statistics. The numbers in parenthesis after ARCH, BP,Q and White refer to the number of degrees of freedom of the χ^2 statistics.

		-2	-1	0	1	2
Δ Long- Δ Short	1-6 mo.	-0.11	0.18	0.70	0.13	-0.03
Yields	1-12 mo.	-0.01	0.13	0.46	0.21	-0.02
	3-6 mo.	-0.08	-0.14	0.95	0.13	-0.10
	3-12 mo.	0.06	0.12	0.76	0.20	-0.01
	6-12 mo.	-0.09	0.11	0.70	0.19	-0.04
$FP-\Delta$ Short	1-3 mo.	-0.08	0.24	-0.29	-0.12	08
Yield	1-6 mo.	-0.05	0.16	-0.25	-0.14	13
	1-12 mo	0.07	0.25	-0.23	-0.08	13
	3-6 mo	-0.07	0.06	-0.14	-0.12	04
	3-12 mo	0.15	0.19	-0.12	-0.11	06
	6-12 mo	0.18	0.18	-0.07	-0.11	02
SP- Δ Short	1-3 mo.	-0.06	0.32	-0.36	-0.06	-0.12
Yield	1-6 mo.	-0.07	0.28	-0.39	-0.08	-0.17
	1-12 mo	0.15	0.34	-0.42	-0.18	-0.22
	3-6 mo	-0.02	0.06	-0.18	-0.11	-0.11
	3-12 mo	0.27	0.24	-0.26	-0.21	-0.13
	6-12 mo	0.34	0.28	-0.23	-0.21	-0.10

Table 3 Cross Moments: 1964,4-1987,11

Note: FP stands for Forward Premium, SP for Spread.

	1 month	3 months	6 months	12 months
	T-Bill	T-Bill	T-Bill	T-Bill
Mean	[-1.12, 7.01]	[3.93, 7.24]	[5.45, 7.42]	[6.61, 7.60]
$P(h(\theta, z_t) < H)$	0.90	0.92	0.93	0.92
Std. Error	[0.43, 37.58]	[0.21, 15.85]	[0.11, 8.03]	[0.06, 4.01]
$P(h(\theta, z_t) < H)$	0.48	0.68	0.79	0.90
Skewness	[-1.49,-0.38]	[-1.47, -0.24]	[-1.46, -0.21]	[-1.45, -0.16]
$P(h(\theta, z_t) < H)$	0.65	0.64	0.72	1.00
Kurtosis	[1.65, 23.66]	[1.36, 23.20]	[1.07, 22.92]	[0.83, 22.81]
$P(h(\theta, z_t) < H)$	0.46	0.43	0.48	0.22
Autocor. 1	[-0.22, 0.65]	[-0.23, 0.73]	[-0.25, 0.75]	[-0.25, 0.76]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00
Autocor. 2	[-0.05, 0.47]	[-0.05, 0.49]	[-0.05, 0.49]	[-0.05, 0.50]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00
Autocor. 4	[-0.05, 0.15]	[-0.06, 0.17]	[-0.06, 0.18]	[-0.06, 0.18]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00
Autocor.12	[-0.07, 0.07]	[-0.08, 0.07]	[-0.09, 0.08]	[-0.09, 0.08]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00
Autocor.24	[-0.08, 0.08]	[-0.09, 0.07]	[-0.09, 0.07]	[-0.10, 0.07]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00
ARCH(13)	[0.06, 45.07]	[0.06, 39.16]	[0.06, 34.87]	[0.06, 34.49]
$P(h(\theta, z_t) < H)$	0.93	0.95	0.95	0.95
BP(13)	[0.27, 37.15]	[0.33, 40.56]	[0.35, 49.43]	[0.32, 49.00]
$P(h(\theta, z_t) < H)$	0.99	0.98	0.98	0.97
White(26)	[2.69, 62.44]	[3.13, 63.02]	[3.27, 69.38]	[3.30, 69.03]
$P(h(\theta, z_t) < H)$	1.00	0.97	0.96	0.97
Q(49)	[7.61, 92.86]	[33.70, 87.16]	[2.25, 97.53]	[5.79, 97.85]
$P(h(\theta, z_t) < H)$	0.61	0.74	0.08	0.34
KS	[0.61, 4.11]	[0.71, 3.98]	[0.96, 3.86]	[0.90, 3.90]
$P(h(\theta, z_t) < H)$	0.36	0.32	0.08	0.26

Table 4 Simulated data 90% Bands: T-Bill Yields

Note: BP refers to Breusch-Pagan test, Q to the Ljung-Box test, KS refers to the Kolmogorov-Smirnov statistics. The numbers in parenthesis after ARCH, BP, White and Q refer to the number of degrees of freedom of the χ^2 statistics.

Table 5 Simulated data 90% Bands: T-Bill Spreads

			Table 5				
		Sir	nulated data				
		90% Ban	ds: T-Bill S	preads			
	1-3 months	1-6 months	1-12 months	3.6 months	3.19 months	6.12 months	=
	1-0 months	1-0 months	1-12 months	5-0 months	5-12 months	0-12 months	
Mean	[0.85, 5.51]	[1.11, 7.05]	[1.44, 8.00]	[0.10, 1.54]	[0.32, 2.62]	[0.17, 1.17]	-
$P(h(\theta, z_t) < H)$	0.00	0.00	0.01	0.09	0.08	0.06	
Std. Error	[0.18, 21.65]	[0.29, 29.58]	[0.36, 33.58]	[0.10, 7.82]	[0.16, 11.84]	[0.05, 4.01]	-
$P(h(\theta, z_t) < H)$	0.17	0.14	0.18	0.26	0.42	0.65	
Skewness	[0.33, 1.49]	[0.32, 1.49]	[0.31, 1.45]	[0.28, 1.48]	[0.26, 1.47]	[0.18, 1.46]	
$\frac{P(h(\theta, z_t) < H)}{V}$	0.18	0.11	0.02	0.08	0.07	0.00	_
Kurtosis $D(L(0, x) \neq H)$	[1.85, 23.69]	[1.76, 23.69]	[1.65, 23.68]	[1.60, 23.47]	[1.46, 23.31]	[0.86, 23.95]	
$\frac{\Gamma(n(\theta, z_t) < H)}{\Lambda uto cor}$	0.51	0.38	0.42	0.10	0.71	0.97	-
$P(h(\theta, z) < H)$	[-0.20, 0.54]	[-0.21, 0.60]	[-0.22, 0.63]	[-0.23, 0.70]	[-0.23, 0.72]	[-0.25, 0.74]	
Autocor 2	0.00	0.82	0.90	0.80	0.93	0.84	- d)
$P(h(\theta \neq) \neq H)$	0.73	0.80	[-0.05, 0.46]	[-0.00, 0.48]	[-0.00, 0.49]	[-0.05, 0.48]	tute
Autocor A	[-0.05.0.15]	[-0.07_0.15]	[0.05 0.15]	0.95	0.90	0.75	- Stil
$P(h(\theta, z_4) < H)$	0 74	0.80	0.33	0.80	[-0.00, 0.10]	[-0.09, 0.17]	-
Autocor 12	[-0.06.0.06]	[-0.08.0.07]	[-0.07 0.07]	[-0.08.0.07]	[0.95	[0.02	-it
$P(h(\theta, z_t) < H)$	1.00	0.98	0.98	0.96	[-0.08, 0.07]	[-0.09, 0.08]	el.
Autocor.24	[-0.08, 0.05]	[-0.08, 0.07]	[-0.08.0.08]	[-0.09.0.08]	[-0.09.0.07]		
$P(h(\theta, z_t) < H)$	1.00	0.99	0.98	0.91	0.14	0.42	\supset
ARCH(13)	[0.06, 45.93]	[0.06, 46.42]	[0.06, 43.98]	[0.06, 44.03]	[0.06, 41.38]	[0.06, 35,59]	- a
$P(h(\theta, z_t) < H)$	0.93	0.98	0.99	0.96	0.97	0.98	be
BP(13)	[0.23, 41.06]	[0.25, 38.98]	[0.24, 38.01]	[0.32, 36.30]	[0.34, 36.00]	[0.36, 48.09]	- Y
$P(h(\theta, z_t) < H)$	0.99	1.00	0.98	0.87	1.00	1.00	Ш.
White(26)	[1.94, 65.37]	[2.22, 67.24]	[2.40, 67.74]	[2.83, 63.07]	[3.01, 63.10]	[3.23, 64.07]	. (S
$P(h(\theta, z_t) < H)$	0.98	1.00	1.00	0.97	1.00	0.99	D
Q(49)	[0.86, 92.78]	[4.04, 89.31]	[0.99, 86.91]	[7.13, 93.68]	[1.74, 80.45]	[2.38, 95.98]	ut
$P(h(\theta, z_t) < H)$	0.67	0.27	0.03	0.76	0.78	0.45	< A
KS	[0.02, 1.68]	[0.66, 2.12]	[0.56, 2.76]	[0.50, 1.56]	[0.57, 1.60]	[0.12, 1.06]	LP.
$P(h(\theta, z_t) < H)$	0.38	0.49	0.57	0.31	0.36	0.64	.0
ote: BP refers to	o Breusch-Pag	an test, Q to t	he Ljung-Box	test, KS refer	s		
to the Ko	lmogorov-Smi	rnov statistics	. The numbers	s in parenthesi	S		
after ARC	CH, BP, Q an	d White refer	to the number	er of degrees o	of		
freedom o	f the χ^2 statis	tics.					
			37				

		-2	-1	0	1	2
Δ Long- Δ Short Yields	1-6 mo.	[-0.17, 0.28]	[-0.64, -0.10]	[0.92, 0.99]	[-0.64, 0.12]	[-0.18, 0.26]
	$P(h(\theta, z_t) < H)$	0.09	1.00	0.00	0.95	0.21
	1-12 mo.	[-0.16, 0.29]	[-0.64, -0.10]	[0.88, 0.99]	[-0.65, 0.13]	[-0.17, 0.27]
	$P(h(\theta, z_t) < H)$	0.25	1.00	0.00	0.96	0.22
	3-6 mo.	[-0.20, 0.26]	[-0.64, 0.03]	[0.994, 0.999]	[-0.64, 0.10]	[-0.20, 0.25]
	$P(h(\theta, z_t) < H)$	0.16	0.98	0.00	0.96	0.15
	3-12 mo.	[-0.20, 0.27]	[-0.64, 0.01]	[0.97, 0.99]	[-0.64, 0.11]	[-0.20, 0.25]
	$P(h(\theta, z_t) < H)$	0.62	0.98	0.00	0.98	0.27
	6-12 mo.	[-0.21, 0.26]	[-0.64, 0.07]	[0.98, 0.99]	[-0.64, 0.09]	[-0.20, 0.25]
	$P(h(\theta, z_t) < H)$	0.17	0.97	0.00	0.98	0.24
$P-\Delta$ Short Yields	1-3 mo.	[-0.22, 0.30]	[0.41, 0.77]	[-0.77, -0.44]	[-0.29, 0.22]	[-0.24, 0.01]
	$P(h(\theta, z_t) < H)$	0.35	0.00	1.00	0.17	0.33
	1-6 mo.	[-0.22, 0.30]	[0.40, 0.77]	[-0.78, -0.43]	[-0.30, 0.22]	[-0.25, 0.01]
	$P(h(\theta, z_t) < H)$	0.42	0.00	1.00	0.16	0.25
	1-12 mo.	[-0.22, 0.30]	[0.40, 0.77]	[-0.78, -0.42]	[-0.30, 0.22]	[-0.25, 0.02]
	$P(h(\theta, z_t) < H)$	0.76	0.00	1.00	0.23	0.26
	3-6 mo.	[-0.21, 0.40]	[0.34, 0.79]	[-0.76, -0.44]	[-0.36, 0.22]	[-0.26, 0.02]
	$P(h(\theta, z_t) < H)$	0.37	0.00	1.00	0.25	0.60
	3-12 mo.	[-0.21, 0.40]	[0.34, 0.79]	[-0.76, -0.43]	[-0.36, 0.22]	[-0.26, 0.02]
	$P(h(\theta, z_t) < H)$	0.78	0.00	1.00	0.25	0.48
	6-12 mo.	[-0.20, 0.40]	[0.31, 0.80]	[-0.76, -0.45]	[-0.37, 0.21]	[-0.26, 0.01]
	$P(h(\theta, z_t) < H)$	0.81	0.00	1.00	0.25	0.74
P- Δ Short Yields	1-3 mo.	[-0.22, 0.29]	[0.41, 0.78]	[-0.75, -0.51]	[-0.26, 0.22]	[-0.23, 0.01]
	$P(h(\theta, z_t) < H)$	0.39	0.02	1.00	0.23	0.26
	1-6 mo.	[-0.22, 0.30]	[0.41, 0.78]	[-0.76, -0.46]	[-0.28, 0.22]	[-0.24, 0.01]
	$P(h(\theta, z_t) < H)$	0.37	0.00	0.99	0.22	0.22
	1-12 mo.	[-0.22, 0.30]	[0.41, 0.77]	[-0.77, -0.43]	[-0.29, 0.22]	[-0.25, 0.01]
	$P(h(\theta, z_t) < H)$	0.85	0.03	0.96	0.12	0.10
	3-6 mo.	[-0.21, 0.40]	[0.35, 0.79]	[-0.77, -0.39]	[-0.38, 0.21]	[-0.27, 0.02]
	$P(h(\theta, z_t) < H)$	0.53	0.00	1.00	0.26	0.31
	3-12 mo.	[-0.20, 0.40]	[0.36, 0.79]	[-0.78, -0.37]	[-0.39, 0.20]	[-0.28, 0.02]
	$P(h(\theta, z_t) < H)$	0.87	0.00	1.00	0.19	0.29
	6-12 mo.	[-0.21, 0.40]	[0.34, 0.79]	[-0.79, -0.35]	[-0.40, 0.21]	[-0.29, 0.02]
	$P(h(\theta, z_{\star}) < H)$	0.88	0.00	1.00	0.21	0.40

	Т	able 6	
	Simul	ated D	ata
0%	bands:	Cross	Moments

90% Bands: T-Bill Yields No Heteroskedasticity in the Exogenous Processes									
No Heteroskedasticity in the Exogenous Processes									
No Heteroskedasticity in the Exogenous Processes									
1 month 3 months 6 months 12 month									
T-Bill T-Bill T-Bill T-Bill									
Mean $[-0.96, 9.75]$ $[3.82, 8.04]$ $[5.80, 8.03]$ $[6.67, 8.43]$									
$P(h(\theta, z_t) < H)$ 0.76 0.77 0.80 0.79									
Std. Error [0.0005, 0.01] [0.002, 0.006] [0.002, 0.003] [0.006, 0.00	5]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Skewness [-0.26, 0.22] [-0.22, 0.13] [-0.11, 0.19] [0.05, 0.11]									
$P(h(\theta, z_t) < H)$ 0.00 0.00 0.00 0.00									
Kurtosis [-0.97, 0.41] [-0.77, 0.06] [-0.84, 0.28] [-0.64, -0.5]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Autocor. 1 [0.02, 0.59] [0.02, 0.49] [-0.03, 0.50] [-0.05, 0.00	2]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Autocor. 2 [0.01, 0.36] [0.06, 0.30] [0.03, 0.28] [0.01, 0.06									
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Autocor. 4 [-0.09, 0.17] [-0.04, 0.12] [-0.04, 0.13] [-0.004, 0.1]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Autocor.12 [-0.10, 0.09] [0.01, 0.17] [-0.03, 0.10] [-0.004, 0.0]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Autocor.24 [-0.12, 0.12] [-0.08, 0.05] [-0.08, 0.08] [0.02, 0.03									
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
ARCH(13) [6.07, 21.47] [5.65, 19.07] [6.37, 18.53] [9.73, 13.9]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
BP(13) [5.14, 18.51] [8.17, 23.69] [5.14, 17.02] [8.35, 12.4]]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
White (26) [14.07, 35.31] [14.07, 35.31] [14.21, 34.23] [19.06, 36.0	5]								
$P(h(\theta, z_t) < H)$ 1.00 1.00 1.00 1.00									
Q(49) [6.87, 58.66] [25.07, 46.91] [2.96, 48.83] [25.39, 32.7)]								
$P(h(\theta, z_t) < H)$ 0.91 1.00 0.06 0.96									
KS [0.67, 2.01] [0.92, 2.88] [0.83, 2.08] [0.67, 2.87									
$P(h(\theta, z_t) < H)$ 0.36 0.18 0.09 0.33									

Table 7

Note: BP refers to Breusch-Pagan test, Q to the Ljung-Box test, KS refers to the Kolmogorov-Smirnov statistics. The numbers in parenthesis after ARCH, BP, White and Q refer to the number of degrees of freedom of the χ^2 statistics.

Tab	le 8
-----	------

Simulated data 90% Bands: T-Bill Spreads No Heteroskedasticity in Exogenous Process

	140 1	leteloskeuast	icity in Exoge	enous i locesa	565	
	1-3 months	1-6 months	1-12 months	3-6 months	3-12 months	6-12 months
Mean	[-1.60 1.62]	[-2.84 2.61]	[-3 39 3 56]	[-1.18.0.99]	[-1.68, 1.94]	-[0.52 0.95]
$P(h(A \sim) < H)$	0.28	0.30	0.25	0.33	1 00	0.20
$\frac{1(n(0, z_t) < 11)}{\text{Std} \text{ Frror}}$		[0 001 0 007]	[0.006_0.01]	[0 001 0 003]	[0 005 0 007]	[0 005 0 005]
$P(h(A, \gamma) < H)$	1 00	1 00	1 00	1 00	1 00	1 00
$\frac{1(n(0,2i) < 11)}{\text{Shown and}}$	[0.20, 0.05]	[0.16.0.20]	[0.06.0.16]	[0.05.0.15]	[0 003 0 12]	[0.05.0.11]
$D(h(\theta \sim) < H)$	[-0.20, 0.03]	[-0.10, 0.20]	1.00	1 00	[0.003, 0.12]	1 00
$\frac{F(n(\theta, z_t) < H)}{V_{\text{contonic}}}$	1.00	1.00	1.00	[0.74 0.11]	[0.75_0.97]	[0.62 0.49]
Rurtosis D(L(0, x) < H)	[-0.91, 0.03]	[-0.03, 0.20]	[-0.05, 0.04]	[-0.74, -0.11]	[-0.75, -0.27]	[-0.03, -0.40]
$\frac{P(n(\theta, z_t) < H)}{1}$	1.00	1.00	1.00	1.00	1.00	1.00
Autocor. 1	[0.0006, 0.43]	[-0.04, 0.51]	[-0.05, 0.32]	[-0.03, 0.40]	[-0.04, 0.17]	[-0.05, 0.01]
$P(h(\theta, z_t) < H)$	0.85	0.77	1.00	1.00	1.00	1.00
Autocor. 2	[0.07, 0.23]	[0.01, 0.28]	[0.008, 0.18]	[0.05, 0.23]	[-0.01, 0.10]	[0.01, 0.04]
$P(h(\theta, z_t) < H)$	0.94	0.97	1.00	1.00	1.00	1.00 +
Autocor. 4	[-0.02, 0.10]	[-0.03, 0.14]	[-0.03, 0.05]	[-0.08, 0.06]	[-0.06, 0.00]	[-0.02, 0.01]
$P(h(\theta, z_t) < H)$	0.88	0.83	0.21	1.00	1.00	1.00 🚊
Autocor.12	[0.01, 0.17]	[-0.06, 0.08]	[-0.04, 0.03]	[0.03, 0.12]	[-0.07, 0.02]	[-0.02, -0.00]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	0.43	1.00	1.00
Autocor.24	[-0.04, 0.06]	[-0.07, 0.09]	[0.004, 0.07]	[-0.04, 0.04]	[0.01, 0.07]	[0.03, 0.06]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	0.96	0.00	0.00
ARCH(13)	[6.37, 17.74]	[6.33, 18.54]	[7.21, 17.74]	[7.70, 17.06]	[10.42, 21.19]	[10.80, 14.92]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00	1.00	1.00 2
BP(13)	[5.58, 17.09]	[5.96, 21.14]	[6.33, 17.06]	[10.95, 27.77]	[7.27, 13.56]	[8.41, 11.51]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	0.61	1.00	1.00
White(26)	[8.12, 31.07]	[7.12, 26.88]	[8.06, 28.13]	[9.54, 29.10]	[11.05, 33.04]	[10.84, 32.28]
$P(h(\theta, z_t) < H)$	1.00	1.00	1.00	1.00	1.00	1.00 4
Q(49)	[22.70, 44.64]	[24.16, 48.87]	[10.60, 46.64]	[28.31, 43.08]	[8.93, 43.94]	[27.45, 37.27]
$P(h(\theta, z_t) < H)$	0.99	0.09	0.02	0.24	1.00	0.98 D
KS	[0.23, 1.60]	[0.80, 1.90]	[0.83, 1.67]	[0.52, 1.80]	[0.73, 1.72]	[0.44, 1.56]
$P(h(\theta, z_t) < H)$	0.28	0.43	0.49	0.36	0.29	0.32 ©

Note: BP refers to Breusch-Pagan test, Q to the Ljung-Box test, KS refers to the Kolmogorov-Smirnov statistics. The numbers in parenthesis after ARCH, BP, Q and White refer to the number of degrees of freedom of the χ^2 statistics.

	90 No Heterosk	0% bands: C edasticity in	the Exogene	ts ous Processes			
		-2	-1	0	1	2	
Δ Long- Δ Short Yields	1-6 mo.	[-0.18, 0.12]	[-0.21, -0.11]	[-0.26, 0.90]	[-0.20, 0.16]	[-0.17, 0.06]	
	$P(h(\theta, z_t) < H)$	0.15	1.00	0.84	0.87	0.43	
	1-12 mo.	[-0.08, 0.13]	[-0.03, 0.26]	[-0.56, 0.12]	[-0.03, 0.31]	[-0.07, 0.10]	
	$P(h(\theta, z_t) < H)$	0.29	0.46	1.00	0.78	0.27	
	3-6 mo.	[-0.11, 0.09]	[-0.43, -0.24]	[0.73, 0.94]	[-0.39, -0.23]	[-0.10, 0.01]	
	$P(h(\theta, z_t) < H)$	0.10	1.00	0.96	1.00	0.10	
	3-12 mo.	[-0.01, 0.08]	[-0.30, -0.15]	[0.34, 0.48]	[-0.26, -0.11]	[-0.06, 0.02]	
	$P(h(\theta, z_t) < H)$	0.45	1.00	1.00	1.00	0.85	
	6-12 mo.	[-0.07, 0.01]	[-0.51, -0.23]	[-0.51, 0.94]	[-0.50, -0.20]	[-0.09, 0.01]	
	$P(h(\theta, z_t) < H)$	0.04	1.00	0.17	1.00	0.75	
FP- Δ Short Yields	1-3 mo.	[0.03, 0.28]	[0.09, 0.49]	[-0.49, -0.10]	[-0.28, 0.03]	[-0.18, 0.04]	
	$P(h(\theta, z_t) < H)$	0.00	0.25	0.73	0.68	0.50	
	1-6 mo.	[0.02, 0.28]	[0.45, 0.57]	[-0.58, -0.44]	[-0.28, 0.02]	[-0.19, 0.07]	
	$P(h(\theta, z_t) < H)$	0.00	0.00	1.00	0.70	0.23	с).
	1-12 mo.	[0.02, 0.26]	[0.33, 0.48]	[-0.49, -0.31]	[-0.25, 0.03]	[-0.17, 0.08]	ħ
	$P(h(\theta, z_t) < H)$	0.39	0.01	1.00	0.53	0.14	E
	3-6 mo.	[-0.01, 0.23]	[0.42, 0.59]	[-0.60, -0.43]	[-0.23, 0.04]	[-0.17, 0.02]	Ē
	$P(h(\theta, z_t) < H)$	0.00	0.00	1.00	0.21	0.89	>
	3-12 mo.	[0.06, 0.22]	[-0.03, 0.38]	[-0.39, 0.02]	[-0.21, 0.007]	[-0.17,-0.04]	S
	$P(h(\theta, z_t) < H)$	0.82	0.82	0.25	0.19	0.93	ē
	6-12 mo.	[-0.05, 0.23]	[-0.64, 0.30]	[-0.27, 0.63]	[-0.23, -0.03]	[-0.18, 0.07]	2
	$P(h(\theta, z_t) < H)$	0.86	0.90	0.14	0.27	1.00	5
SP- Δ Short Yields	1-3 mo.	[-0.00, 0.22]	[-0.18, 0.41]	[-0.41, 0.16]	[-0.25, 0.05]	[-0.14, 0.04]	E
	$P(h(\theta, z_t) < H)$	0.00	0.85	0.13	0.44	0.11	ĕ
	1-6 mo.	[0.01, 0.27]	[0.36, 0.49]	[-0.51, -0.35]	[-0.26, 0.03]	[-0.18, 0.07]	0
	$P(h(\theta, z_t) < H)$	0.00	0.01	0.79	0.70	0.07	ħ
	1-12 mo.	[0.03, 0.20]	[0.23, 0.45]	[-0.45, -0.20]	[-0.21, 0.03]	[-0.11, 0.08]	Ц.
	$P(h(\theta, z_t) < H)$	0.87	0.53	0.14	0.09	0.00	S
	3-6 mo.	[-0.01, 0.19]	[0.50, 0.62]	[-0.62, -0.50]	[-0.17, 0.04]	[-0.14, -0.03]) JO
	$P(h(\theta, z_t) < H)$	0.04	0.00	1.00	0.14	0.12	Ĕ
	3-12 mo.	[0.04, 0.18]	[-0.09, 0.24]	[-0.23, 0.08]	[-0.15, 0.01]	[-0.13,-0.04]	2
	$P(h(\theta, z_t) < H)$	1.00	0.94	0.05	0.00	0.06	ð
	6-12 mo.	[0.01, 0.12]	[-0.66, -0.09]	[0.11, 0.66]	[-0.10,-0.01]	[-0.14,-0.05]	Ĩ
	$P(h(\theta, z_t) < H)$	1.00	1.00	0.00	0.00	0.76	0

 Table 9

 Simulated Data

 90% bands: Cross Moments

 No Heteroskedasticity in the Exogenous Processe

Note: SP stands for Spread and FP for Forward Premium.

Digitised version produced by the EUI Library in 2020. Available Open Access on Cadmus, European University Institute Research Repository.



Figure 1



Figure 2

Digitised version produced by the EUI Library in 2020. Available Open Access on Cadmus, European University Institute Research Repository.

© The Author(s). European University Institute. Digitised version produced by the EUI Library in 2020. Available Open Access on Cadmus, European University Institute Research Repository.

Working Papers of the Department of Economics Published since 1989

89/370

B. BENSAID/ R.J. GARY-BOBO/ S. FEDERBUSCH The Strategic Aspects of Profit Sharing in the Industry

89/374 Francisco S. TORRES Small Countries and Exogenous Policy Shocks

89/375 Renzo DAVIDDI Rouble Convertibility: A Realistic Target

89/377 Elettra AGLIARDI On the Robustness of Contestability Theory

89/378 Stephen MARTIN The Welfare Consequences of Transaction Costs in Financial Markets

89/381 Susan SENIOR NELLO Recent Developments in Relations Between the EC and Eastern Europe

89/382 Jean GABSZEWICZ/ Paolo GARELLA/ Charles NOLLET Spatial Price Competition With Uninformed Buyers

89/383 Benedetto GUI Beneficiary and Dominant Roles in Organizations: The Case of Nonprofits

89/384 Agustín MARAVALL/ Daniel PEÑA Missing Observations, Additive Outliers and Inverse Autocorrelation Function

89/385 Stephen MARTIN Product Differentiation and Market Performance in Oligopoly

89/386 Dalia MARIN Is the Export-Led Growth Hypothesis Valid for Industrialized Countries? 89/387 Stephen MARTIN

Modeling Oligopolistic Interaction

89/388 Jean-Claude CHOURAQUI The Conduct of Monetary Policy: What have we Learned From Recent Experience

89/390 Corrado BENASSI Imperfect Information and Financial Markets: A General Equilibrium Model

89/394

Serge-Christophe KOLM Adequacy, Equity and Fundamental Dominance: Unanimous and Comparable Allocations in Rational Social Choice, with Applications to Marriage and Wages

89/395

Daniel HEYMANN/ Axel LEIJONHUFVUD On the Use of Currency Reform in Inflation Stabilization

89/400

Robert J. GARY-BOBO On the Existence of Equilibrium Configurations in a Class of Asymmetric Market Entry Games *

89/402 Stephen MARTIN Direct Foreign Investment in The United States

89/413 Francisco S. TORRES Portugal, the EMS and 1992: Stabilization and Liberalization

89/416 Joerg MAYER Reserve Switches and Exchange-Rate Variability: The Presumed Inherent Instability of the Multiple Reserve-Currency System

89/417

José P. ESPERANÇA/ Neil KAY Foreign Direct Investment and Competition in the Advertising Sector: The Italian Case

* Working Paper out of print

89/418 Luigi BRIGHI/ Mario FORNI Aggregation Across Agents in Demand Systems

89/420 Corrado BENASSI A Competitive Model of Credit Intermediation

89/422 Marcus MILLER/ Mark SALMON When does Coordination pay?

89/423 Marcus MILLER/ Mark SALMON/ Alan SUTHERLAND Time Consistency, Discounting and the Returns to Cooperation

89/424 Frank CRITCHLEY/ Paul MARRIOTT/ Mark SALMON On the Differential Geometry of the Wald Test with Nonlinear Restrictions

89/425 Peter J. HAMMOND On the Impossibility of Perfect Capital Markets

89/426 Peter J. HAMMOND Perfected Option Markets in Economies with Adverse Selection

89/427 Peter J. HAMMOND Irreducibility, Resource Relatedness, and Survival with Individual Non-Convexities

* * *

ECO No. 90/1** Tamer BAŞAR and Mark SALMON Credibility and the Value of Information Transmission in a Model of Monetary Policy and Inflation

ECO No. 90/2 Horst UNGERER The EMS – The First Ten Years Policies – Developments – Evolution

ECO No. 90/3 Peter J. HAMMOND Interpersonal Comparisons of Utility: Why and how they are and should be made ECO No. 90/4 Peter J. HAMMOND A Revelation Principle for (Boundedly) Bayesian Rationalizable Strategies

ECO No. 90/5 Peter J. HAMMOND Independence of Irrelevant Interpersonal Comparisons

ECO No. 90/6 Hal R. VARIAN A Solution to the Problem of Externalities and Public Goods when Agents are Well-Informed

ECO No. 90/7 Hal R. VARIAN Sequential Provision of Public Goods

ECO No. 90/8 T. BRIANZA, L. PHLIPS and J.F. RICHARD Futures Markets, Speculation and Monopoly Pricing

ECO No. 90/9 Anthony B. ATKINSON/ John MICKLEWRIGHT Unemployment Compensation and Labour Market Transition: A Critical Review

ECO No. 90/10 Peter J. HAMMOND The Role of Information in Economics

ECO No. 90/11 Nicos M. CHRISTODOULAKIS Debt Dynamics in a Small Open Economy

ECO No. 90/12 Stephen C. SMITH On the Economic Rationale for Codetermination Law

ECO No. 90/13 Elettra AGLIARDI Learning by Doing and Market Structures

ECO No. 90/14 Peter J. HAMMOND Intertemporal Objectives

ECO No. 90/15 Andrew EVANS/Stephen MARTIN Socially Acceptable Distortion of Competition: EC Policy on State Aid

^{**} Please note: As from January 1990, the EUI Working Papers Series is divided into six sub-series, each series will be numbered individually (e.g. EUI Working Paper LAW No. 90/1).

ECO No. 90/16 Stephen MARTIN Fringe Size and Cartel Stability

ECO No. 90/17 John MICKLEWRIGHT Why Do Less Than a Quarter of the Unemployed in Britain Receive Unemployment Insurance?

ECO No. 90/18 Mrudula A. PATEL Optimal Life Cycle Saving With Borrowing Constraints: A Graphical Solution

ECO No. 90/19 Peter J. HAMMOND Money Metric Measures of Individual and Social Welfare Allowing for Environmental Externalities

ECO No. 90/20 Louis PHLIPS/ Ronald M. HARSTAD Oligopolistic Manipulation of Spot Markets and the Timing of Futures Market Speculation

ECO No. 90/21 Christian DUSTMANN Earnings Adjustment of Temporary Migrants

ECO No. 90/22 John MICKLEWRIGHT The Reform of Unemployment Compensation: Choices for East and West

ECO No. 90/23 Joerg MAYER U. S. Dollar and Deutschmark as Reserve Assets

ECO No. 90/24 Sheila MARNIE Labour Market Reform in the USSR: Fact or Fiction?

ECO No. 90/25 Peter JENSEN/ Niels WESTERGÅRD-NIELSEN Temporary Layoffs and the Duration of Unemployment: An Empirical Analysis

ECO No. 90/26 Stephan L. KALB Market-Led Approaches to European Monetary Union in the Light of a Legal Restrictions Theory of Money ECO No. 90/27 Robert J. WALDMANN Implausible Results or Implausible Data? Anomalies in the Construction of Value Added Data and Implications for Estimates of Price-Cost Markups

ECO No. 90/28 Stephen MARTIN Periodic Model Changes in Oligopoly

ECO No. 90/29 Nicos CHRISTODOULAKIS/ Martin WEALE Imperfect Competition in an Open Economy

* * *

ECO No. 91/30 Steve ALPERN/Dennis J. SNOWER Unemployment Through 'Learning From Experience'

ECO No. 91/31 David M. PRESCOTT/Thanasis STENGOS Testing for Forecastible Nonlinear Dependence in Weekly Gold Rates of Return

ECO No. 91/32 Peter J. HAMMOND Harsanyi's Utilifarian Theorem: A Simpler Proof and Some Ethical Connotations

ECO No. 91/33 Anthony B. ATKINSON/ John MICKLEWRIGHT Economic Transformation in Eastern Europe and the Distribution of Income

ECO No. 91/34 Svend ALBAEK On Nash and Stackelberg Equilibria when Costs are Private Information

ECO No. 91/35 Stephen MARTIN Private and Social Incentives to Form R & D Joint Ventures

ECO No. 91/36 Louis PHLIPS Manipulation of Crude Oil Futures

ECO No. 91/37 Xavier CALSAMIGLIA/Alan KIRMAN A Unique Informationally Efficient and Decentralized Mechanism With Fair Outcomes

* Working Paper out of print

ECO No. 91/38 George S. ALOGOSKOUFIS/ Thanasis STENGOS Testing for Nonlinear Dynamics in Historical Unemployment Series

ECO No. 91/39 Peter J. HAMMOND The Moral Status of Profits and Other Rewards: A Perspective From Modern Welfare Economics

ECO No. 91/40 Vincent BROUSSEAU/Alan KIRMAN The Dynamics of Learning in Mis-Specified Models

ECO No. 91/41 Robert James WALDMANN Assessing the Relative Sizes of Industry- and Nation Specific Shocks to Output

ECO No. 91/42 Thorsten HENS/Alan KIRMAN/Louis PHLIPS Exchange Rates and Oligopoly

ECO No. 91/43 Peter J. HAMMOND Consequentialist Decision Theory and Utilitarian Ethics

ECO No. 91/44 Stephen MARTIN Endogenous Firm Efficiency in a Cournot Principal-Agent Model

ECO No. 91/45 Svend ALBAEK Upstream or Downstream Information Sharing?

ECO No. 91/46 Thomas H. McCURDY/ Thanasis STENGOS A Comparison of Risk-Premium Forecasts Implied by Parametric Versus Nonparametric Conditional Mean Estimators

ECO No. 91/47 Christian DUSTMANN Temporary Migration and the Investment into Human Capital

ECO No. 91/48 Jean-Daniel GUIGOU Should Bankruptcy Proceedings be Initiated by a Mixed Creditor/Shareholder?

ECO No. 91/49 Nick VRIEND Market-Making and Decentralized Trade ECO No. 91/50 Jeffrey L. COLES/Peter J. HAMMOND Walrasian Equilibrium without Survival: Existence, Efficiency, and Remedial Policy

ECO No. 91/51 Frank CRITCHLEY/Paul MARRIOTT/ Mark SALMON Preferred Point Geometry and Statistical Manifolds

ECO No. 91/52 Costanza TORRICELLI The Influence of Futures on Spot Price Volatility in a Model for a Storable Commodity

ECO No. 91/53 Frank CRITCHLEY/Paul MARRIOTT/ Mark SALMON Preferred Point Geometry and the Local Differential Geometry of the Kullback-Leibler Divergence

ECO No. 91/54 Peter MØLLGAARD/ Louis PHLIPS Oil Futures and Strategic Stocks at Sea

ECO No. 91/55 Christian DUSTMANN/ John MICKLEWRIGHT Benefits, Incentives and Uncertainty

ECO No. 91/56 John MICKLEWRIGHT/ Gianna GIANNELLI Why do Women Married to Unemployed Men have Low Participation Rates?

ECO No. 91/57 John MICKLEWRIGHT Income Support for the Unemployed in Hungary

ECO No. 91/58 Fabio CANOVA Detrending and Business Cycle Facts

ECO No. 91/59 Fabio CANOVA/ Jane MARRINAN Reconciling the Term Structure of Interest Rates with the Consumption Based ICAP Model

© The Author(s). European University Institute. Digitised version produced by the EUI Library in 2020. Available Open Access on Cadmus, European University Institute Research Repository. 2



© The Author(s). European University Institute. Digitised version produced by the EUI Library in 2020. Available Open Access on Cadmus, European University Institute Research Repository.



